

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bagnato, Luca

Working Paper From taxation to fighting for the nation: Historical fiscal capacity and military draft evasion during WWI

Working Paper, No. 423

Provided in Cooperation with: Department of Economics, University of Zurich

Suggested Citation: Bagnato, Luca (2024) : From taxation to fighting for the nation: Historical fiscal capacity and military draft evasion during WWI, Working Paper, No. 423, University of Zurich, Department of Economics, Zurich, https://doi.org/10.5167/uzh-224846

This Version is available at: https://hdl.handle.net/10419/289564

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



University of Zurich

Department of Economics

Working Paper Series

ISSN 1664-7041 (print) ISSN 1664-705X (online)

Working Paper No. 423

From Taxation to Fighting for the Nation: Historical Fiscal Capacity and Military Draft Evasion during WWI

Luca Bagnato

Revised version, March 2024

From Taxation to Fighting for the Nation: Historical Fiscal Capacity and Military Draft Evasion during WWI

Luca Bagnato*

Abstract

Do strong states affect the culture and actions of their citizens in a persistent way? And if so, can the capacity to tax, by itself, drive this effect? I study how the historical capacity of a state to collect taxes affects the decision of citizens to evade the mandatory military draft. I look at Italy during World War I and identify quasi-exogenous variation in tax collection induced by the administrative structure of Piedmont during the 1814-1870 period. Using newly collected individual data on the universe of the 1899 cohort drafted in the province of Turin, I find that citizens born in towns with lower historical fiscal capacity are more likely to evade the military draft, and that the effect transmits through changes in culture. Placebo estimates confirm that the effect can be attributed to fiscal capacity and is not confounded by legal capacity. Additional results on voter turnout support the interpretation that higher fiscal capacity led citizens to perceive higher state capacity and, consequently, higher returns to participation.

Keywords: Fiscal Capacity, Tax Collection, Culture, Military Draft, World War I **JEL classification:** D70; D73; D74; D91; H20; N4; P0

^{*}University of Milan, Department of Economics, Management, and Quantitative Methods, Milan, Italy. Email: luca.bagnato@unimi.it. I am indebted to David Yanagizawa-Drott, Pietro Biroli, Lorenzo Casaburi, and Bruno Caprettini for invaluable support, advice, and encouragement at all stages of this project. I am also grateful to Michelle Acampora, Elliott Ash, Samuel Bazzi, Dan Bogart, Cédric Chambru, Ana Costa-Ramón, Luisa Dolza, Giovanni Facchini, Gloria Gennaro, Michela Giorcelli, Paola Giuliano, Andrea Hofer, Leander Heldring, Gianni Marciante, Nathan Nunn, Dina Pomeranz, Alessandro Saia, Francesca Salvati, Lukas Schmid, Giuseppe Sorrenti, Marco Tabellini, Miriam Venturini, Joachim Voth, Romain Wacziarg, David Yang and Ulf Zölitz for very helpful conversations. I would like to thank discussants and seminar participants at the UZH Empirical Doctoral Micro Seminar, the Zurich Political Economy Seminar Series, the Swiss Workshop on Political Economy and Development, the CSQIEP Virtual Seminar, the University of Milan, Sabanci University, UC Irvine, the UCLA Anderson GEM Seminar, the UCLA Economic History Proseminar, the 14th EHES Conference, the 2022 EWMES, the 2023 EHS Annual Conference, and the ASREC Europe 2023 Conference for valuable comments and feedback. I also thank Giulia Caccia and Giampiero Sica for their support during the data collection, and Giovanni Facchini for sharing his data. Giuseppe Allegra provided excellent research assistance. I gratefully acknowledge financial support by the Swiss National Science Foundation through the SNF Doc.Mobility (grant P1ZHP1_187859) and by the UZH Graduate Campus through a Travel Grant.

1 Introduction

The ability of a state's institutions to enforce laws and collect taxes efficiently (namely, *state capacity*) is a fundamental driver of economic development (Kaldor, 1963; Besley and Persson, 2009, 2010; Acemoglu et al., 2015; Dell et al., 2018; Dincecco et al., 2021).

State capacity does not only affect the economic situation of its citizens, but impacts citizens' culture as well (Weber, 1976; Becker et al., 2016; Dell et al., 2018; Lowes et al., 2017; Heldring, 2021). Norms developed under strong states can persist over time, and societies exposed to strong state capacity may show the effects of such exposure in the actions of their citizens even generations after the establishment (and the removal) of the institutions that make a state strong.

In order to build state capacity, governments need to collect taxes effectively (Pomeranz and Vila-Belda, 2019). Historically, for most citizens, taxation was the first activity through which the state started to be present in their lives (Cantoni et al., 2019), which makes the ability to collect taxes a hallmark of the state (Scott, 2017). Recent findings also suggest that the act of enforcing taxation on citizens can affect their attitudes towards the government in the short run (Weigel, 2020): thereby, improving the capacity of a government to raise taxes (i.e., *fiscal capacity*) may have important effects on citizens' culture and their actions, potentially also in the long run. However, despite the relevance of fiscal capacity, assessing its effects causally remains a demanding task, as higher levels of fiscal capacity are typically associated with other phenomena, such as high legal capacity.¹²

In this paper, I use individual data on military enlistment and draft evasion during World War I to identify how historical exposure of local communities to fiscal capacity affects citizens' actions in the long run, and in particular, actions that involve a duty towards the state. I look at the Italian military draft in 1917, when the Italian Kingdom was involved in a demanding war and its male population was subject to mass mobilisation to fight through mandatory military service for every Italian man able to serve.

In my analysis I investigate the effect that the historical capacity of a state to collect taxes from its territories had on the norms of local communities, which may have persisted in the long run and ended up affecting citizens' decision to enlist or to evade the military draft. I do this by analysing the long run effects of exposure to institutions for tax collection in place during the 1814-1870 period, a critical juncture for investments in state- and nation-building in Europe. I focus on the Sardinia Kingdom, an independent state that, after experiencing institutional innovation under the Napoleonic domination, and specularly to other European countries at the time, underwent a remarkable state capacity-building process that eventually

¹Legal capacity is intended as the overall ability of a state to enforce laws. The association between legal and fiscal capacity happens because investments in the two are complements (Besley and Persson, 2009) and most determinants of fiscal capacity lead to higher legal capacity too.

²It is also common to use information of tax revenues of countries (or other administrations) as a measure of state capacity at large, abstracting from consideration on its legal and fiscal components (Gennaioli and Voth, 2015; Johnson, 2015; Heldring, 2023).

led it to evolve into the Kingdom of Italy in 1861.³ The administrative structure of the Sardinia Kingdom provides a unique setting to isolate the effect of fiscal capacity, and to identify variation in it at a quite fine-grained level. In particular, given the partition of the Sardinia Kingdom into small tax districts and the allocation of tax collectors by the government, I leverage a town's distance from tax collectors to identify variation in the capacity of the government to raise taxes from the different towns of the Kingdom. Between 1814 and 1870 every tax district of the Kingdom was assigned to one tax collector only, who was required to live and work in a specific town of the district chosen by the government (the tax district capital) and to visit each town of his district at least once a month. While tax collectors respected the first requirement, anecdotal evidence suggests the second rule was not enforced, so that tax collectors did not move around their tax districts in a systematic way. In 1870 this tax collection system was replaced by a new one under which tax collection was assigned to private citizens separately for each town, so that the previously existing tax districts did not have any role for tax collection in 1917. By studying this setting, I analyse the long run effect on citizens' actions of exposure to institutions granting higher fiscal capacity *after* they were removed (i.e., an *historical* effect).

Motivated by the evidence on tax collectors' behaviour, I use the distance of a town from the residence of its tax collector as a measure of the capacity of the government to collect taxes in it in the 1814-1870 period. While distance from governmental offices is a reasonable and widely used measure of their capacity to act (Michalopoulos and Papaioannou, 2014; Restrepo, 2015; Henn, 2022), in this setting the estimation of the (historical) fiscal capacity effect poses important identification challenges. First, being district capitals, towns hosting tax collectors tend to be larger and richer: therefore, distances from former residences of tax collectors are likely to correlate with several socio-economic characteristics of towns and of citizens living in these towns (e.g. income, market access, social structures), which may in turn drive the decision to evade the military draft. Second, as tax districts were centers of administrative divisions (albeit very small ones), they sometimes hosted also other public offices: thus, distance from a former tax collector's residence could predict not only historical fiscal capacity, but also other components of state capacity. In order to address the first threat to identification I exploit discontinuities in distance from a historical tax collector's residence created by tax district borders. More specifically, I focus on variation in distance from tax collectors' residences within *pairs* of towns that are neighbours, but belonged to two different (and bordering) tax districts before 1870. Neighbouring towns, because of their geographical proximity, have similar observable characteristics, but because of their assignment to two different tax districts, they differ in distance from their former tax collectors' residences. More specifically, I match each town on the border of a tax district to every neighbouring town on the other side of the border and estimate the effect of distance from a former tax collector's residence on draft evasion controlling for a full set of town-pairs

³The political history of the Sardinia Kingdom makes it also directly comparable to the German case, where the strongest independent state (Prussia) unified the country in the second half of the XIX Century.

fixed effects (following Dube et al., 2010). In this way, the effect of historical fiscal capacity on the likelihood of evading the draft is identified by exploiting variation in distance from a tax collector's residence between neighbouring towns (or citizens born in two neighbouring towns) belonging to different tax districts. To address the second challenge, namely the presence of other public offices, I adopt three strategies. First, I exclude from the sample towns for which the former tax collector's residence also hosted a division capital or the closest police station. Second, I verify that no other public office was located one-to-one in tax district capitals, except minor courts. Third, I use a placebo test to disentangle the effect of historical fiscal capacity effect from the one of proximity to a minor court.

To study my research question, I assemble a unique dataset combining newly collected and digitised individual data on all the men from the 1899 cohort drafted in the Province of Turin and town-level information on a large array of both pre- and post-1870 characteristics. The dataset includes rich information on drafted Italian citizens coming from their medical examination and the record of their history in the military until the moment of actual enlistment. Crucially, the dataset contains information on a citizen's name, town of birth, town of residence, and whether they enlisted or evaded the military draft. The individuallevel dataset includes 92-98% of all the men estimated to have been born in 1899 in the Province of Turin and to have survived until the age of 18 (the age of the draft), reaching nearly-full coverage: accordingly, the estimated results are likely to be representative of the male population born in the towns of the sample. Selecting the 1899 cohort allows to study the draft evasion decision of men drafted exactly during World War I, in the moment of utmost effort of the Italian Army. Turin (located in Piedmont, in Northwestern Italy) was the Italian province with the most towns in 1917, and its choice makes it feasible to collect data on thousands of individuals from hundreds of birthplaces in a single archive.

In a first result, I provide evidence that the distance of a town from a tax collector's residence did lead to less taxes collected in that town before 1870, looking at taxes raised by the tax collector on behalf of the town in the divisions of Ivrea and Saluzzo. These findings support the strategy of using distance from a tax collector as a valid measure of (historical) fiscal capacity.

In my main result, I show that historical fiscal capacity lowers the likelihood of citizens to evade the military draft during World War I: being born in a town farther from the former residence of a tax collector predicts a higher likelihood to evade the military draft in 1917, both in town-level and individual-level analyses. Depending on the specification, a 1-kilometre increase in distance from a tax collector's residence leads to 0.2-0.49 percentage points increase in the share of draft evaders in the town; this corresponds to 4-10% of the baseline evasion rate of 4,72%. This positive result is robust to controlling for a large array of town and individual characteristics, holds also under different transformations of the independent and of the dependent variable, and high values of Oster (2019)'s δ suggest unobservables are unlikely to drive the result. To further attenuate concerns regarding a confounding effect of differences in draft enforcement, I also show that, when estimating the effect of distance from a former tax collector's residence and the one of distance from a police station jointly, the historical fiscal capacity effect is basically unaffected in its magnitude and significance. This result, together with the sample selection I impose, shows that the effect I identify is not driven by towns more difficult to tax being also more difficult to patrol. Despite arising from a sample of relatively small towns located in Northwestern Italy, my results are comparable in sign and magnitude to findings from previous research that studied, in many different settings, the interaction between state capacity and culture, the drivers of draft evasion, and the effects of taxation in the XIX Century.

The relationship between distance from a former tax collector's residence and draft evasion during World War I may conflate both the historical fiscal capacity effect and a (historical and contemporary) legal capacity effect, because of the one-to-one correspondence between tax collectors' residences and the location of minor courts. Taking advantage of the administrative system of the unified Italian Kingdom, I run a placebo test using distances from judicial district capitals that never hosted the residence of a tax collector. Using data from the province of Vicenza, I find scarce support for the hypothesis that proximity to a minor court had an effect on the decision to evade the military draft, or to enlist: estimated coefficients are small, often negative, not significant, and unstable across specifications.

Next, I study channels of transmission, exploring whether the effect of historical fiscal capacity is transmitted through culture (affecting the norms of communities more or less exposed to fiscal capacity), or rather through persistent differences in the strength of the state's institutions. I repeat the analysis focusing only on the sample of internal migrants (citizens living in a town different from the one they were born in) and including town-ofresidence fixed effects, keeping fixed every current institutional factor that may affect the decision to evade the military draft. Intuitively, for migrants, variation in the distance of their town of birth from a former tax collector's residence predicts a cultural component in the effect of historical fiscal capacity. I find that distance of the town of birth of a citizen from the former residence of a tax collector has a positive effect on the likelihood of a citizen to evade the draft, with coefficients of distance larger in size compared to the ones estimated on the pooled sample that includes also natives. On the contrary, both the result on the effect of proximity to the police, the placebo test and additional analyses on persistence in local fiscal capacity make institutions unlikely to be the major channel through which the historical fiscal capacity effect transmits in time. Instead, the findings corroborate the hypothesis that historical fiscal capacity transmits its effect in the long run through culture.

After identifying culture as the most relevant channel of transmission, I explore which phenomenon can rationalise the nexus between fiscal capacity and military enlistment, and explain its persistence in time. Motivated both by the interpretation of military enlistment as a citizen's involvement with the state's activity and by recent evidence on citizens' reaction to the introduction of tax collection (Weigel, 2020) I test for the effect of fiscal capacity on political participation in the XIX and the XX Centuries, by looking at voter turnout in 1865 and 1913. I find that, both when the tax system under analysis was in place (1865) and in the year before the unexpected outbreak of World War I (1913), voter turnout was consistently higher in towns closer to their (former) tax collector's residence. This result is consistent with fiscal capacity (through the activity of the tax collector) sending a signal of higher state capacity, leading citizens to update positively their perception of what governments are able to do, and increasing their willingness to influence governments' decisions and participate in their activity. This works as a mechanism for lower draft evasion, as in this setting, military enlistment can arise both as another form of participation in the state's activities and as an action to defend the investment made in the independent state, either in terms of taxes paid and of effort put in its political activity. The main result of this paper can thus be explained by higher perceived returns to involvement with the state.

Another potential mechanism is public good provision. Intuitively better tax collection could have allowed towns to fund more public goods, and this may have, in turn, induced in citizens better attitudes (or even gratitude) towards the state. However, despite being in a setting where taxes were reinvested, looking at the effects on several indicators of public spending I find no evidence that public good provision was higher in towns where fiscal capacity was higher, despite some differences in local-level tax revenues even after 1870; consequently, the provision of some particularly salient public goods is unlikely to drive the main effect. I also provide evidence that proximity to tax collectors did not increase the patriotism of citizens: while state presence (signalled by a tax collector) may have induced citizens to identify more with the central state, I find no consistent effects of distance from a former tax collector's residence on Medals of Honor during World War I or share of citizens named after a Savoy king. The main result of this paper could also be explained by two other mechanisms. The first one is perceived fairness of taxation: more systematic tax collection may have been perceived fairer by citizens (Weigel and Kabue Ngindu, 2023), leading them to recognise higher legitimacy to the state's requests. While plausible in principle, such a mechanism is less likely to be at play in this specific setting, where tax collectors had no power in determining the tax base and its fairness. Finally, a plausible mechanism driving my findings is the emergence of norms of rule-following as a result of exposure to higher tax collection. Citizens living in towns where fiscal capacity was higher may have gotten accustomed to paying taxes, favouring the emergence of norms of tax compliance that could have, in turn, fostered a persistent, generalised norm of rule-following, inducing citizens to comply with the law also in domains different from tax compliance (a spillover effect in line with Keizer et al., 2008). While being consistent with my empirical analysis, data limitations prevent me from testing such a mechanism, which is nonetheless not an alternative to higher perceived returns from involvement with the state.

Ultimately, while the emergence of taxation-induced norms or rule-following remains as a possible explanation, the findings from this paper are mostly consistent with the first described mechanism, i.e. higher perceived state capacity and returns to participation in towns where the state displayed higher fiscal capacity.

This paper mainly contributes to four strands of literature. First, it relates to the litera-

ture on the consequences of establishing state capacity (Dincecco and Katz, 2014; Dincecco et al., 2021; Cantoni et al., 2019) and more specifically on literature studying the impact of state capacity on the attitudes and actions of their citizens (Weber, 1976; Elias, 1994; Pinker, 2011; Becker et al., 2016; Lowes et al., 2017; Johnson, 2015; Heldring, 2021, 2023), which found mixed evidence on how strong states affect trust, rule-following, and nation building in the long run. While these studies looked at effects of state capacity that arise several generations after the institutions under study were discontinued, the main contribution of my paper is to complement this body of literature focusing on the effect of *fiscal* capacity on culture and actions, and isolating it from other components of state capacity like legal capacity and military enforcement.

Second, this paper also contributes to the literature analysing how taxation and tax collection influence the intrinsic motivation to pay taxes, social unrest, political participation, and citizens' perception of state capacity (Levi, 1996; Besley et al., 2021; Lax-Martinez et al., 2022; Weigel, 2020; Weigel and Kabue Ngindu, 2023). I contribute to this literature by analysing a case related more to differences in the effectiveness of the state in raising taxes than to changes in citizens' tax burden, by focusing on the long run cultural effect of imposing fiscal capacity on a community, and by studying a phenomenon other than tax morale or political participation, namely law-abiding behaviour when obligations towards the state are involved.

Third, this paper speaks to the literature on the drivers of participation in war and the determinants of patriotism. A branch of research analysed how socio-economic environmental factors (Kleykamp, 2006), shared experiences (Ronconi and Ramos-Toro, 2023), and economic policies of the state (Alesina et al., 2020, 2021; Caprettini and Voth, 2023) can induce citizens to join the army, e.g. as a reaction to higher public good provision or welfare spending. Another strand of this literature studies the cultural determinants of war service and patriotism (Costa and Kahn, 2003; Campante and Yanagizawa-Drott, 2015; Chen, 2016; Qian and Tabellini, 2021; Esposito et al., 2023), highlighting how participation in war and desertion can be significantly affected by a company's characteristics, cultural transmission from fathers, deterrence of death penalty, perceived state legitimacy, experiences of discrimination, and racist and revisionist narratives about civil wars. I add to this literature by investigating the role of a novel driver in the decision to enlist or evade the draft, namely the long run effect of past fiscal capacity experienced by a community, an arguably powerful signal of the presence of the state; I also focus on a setting where the military draft was mandatory, therefore looking at a decision involving both patriotic considerations and obedience to the law.

Finally, this paper is related to research on mobilization for war (see e.g. Rouanet and Piano, 2023) and for World War I in particular (Koenig, 2023; Berg et al., 2019; Acemoglu et al., 2022; Boehnke and Gay, 2022), which often studied the effects of participation in war on various social and political outcomes. Similarly to Esposito et al. (2023), I contribute to this strand of literature focusing on a novel determinant of participation in World War I.

The rest of this paper is organised as follows: section 2 provides an overview of the tax collection system in Piedmont before 1870 and of the Italian military draft in 1917; section 3 describes the data used in the empirical analysis; section 4 presents the identification strategy; section 5 shows balancedness in observable characteristics between more and less intensely treated units; section 6 presents the main empirical results; section 7 studies mechanisms and section 8 concludes.

2 Background

2.1 Tax collection in the Sardinia Kingdom, 1814-1870

The Sardinia Kingdom was one of the eight most relevant political units of the Italian territories in the first half of the XIX century, i.e. after the Napoleonic wars and before unification. Its territories mainly coincided with four Italian regions, namely Piedmont, Liguria, Aosta Valley and the Sardinian Island.

Except for the years under French domination,⁴ the Sardinia Kingdom (or Piedmont) was a long-living independent state dominated by the House of Savoy. Starting in 1814, the kingdom underwent several political and economic changes, also as a consequence of the revolutions of 1848 and the Italian Wars of Independence, culminating with the unification and the establishment of the Italian Kingdom in 1861, a process that the Sardinia Kingdom led.

The kingdom had a rather complex geographical administrative structure, which it partially inherited from the policies promoted by the French rulers. In particular, after coming back to power in 1814, the House of Savoy kept in place most of the judicial system established by the French, including the partition of Piedmont's territories into a large number of small judicial districts.

The judicial districts were established according to a French law⁵ requiring them to have around 10.000 citizens living in each of them and an area of around 250 square kilometres. In addition, the law required the judge (a Justice of Peace) to live in a designated town of the district, chosen according to its relevance and its centrality with respect to the other towns. Once reinstated in power, the Piedmontese government expanded this system to all its other territories; furthermore, starting in 1814, the judicial districts were also used for tax purposes, assigning one tax collector to each judicial/tax district, and requiring him to live in the same city of the judge.⁶ This administrative division was relevant only for tax and judicial purposes; there was no other public office located in the district capital on a systematic basis, and the district did not have a local government.

⁴The mainland regions of the Sardinia Kingdom were completely annexed to France between 1802 and 1814.

⁵ Règles générales sur l'administration de la Justice et sur l'organisation des Tribunaux dans le Piémont (1801), shown in Figure C.1.

⁶The overlap of judicial and tax districts poses concerns to the identification of the effect of historical fiscal capacity on military draft evasion; in section 4.3 I propose a strategy to address this issue.

Through this system, the collection of direct taxes was centralised and internalised. Tax collectors were public employees, and they were entrusted with the task of collecting both all direct taxes for the central and provincial governments, and every direct or indirect local tax the town government decided to levy.⁷

Piedmont was the Italian pre-unitary state levying more taxes in the first half of the XIX Century (Dincecco et al., 2011); during this period, direct taxes grew of importance for the Sardinia Kingdom in XIX century, almost doubling between 1825 and 1859.⁸ Direct taxes consisted mainly of a land tax and a house tax on owned buildings (Norsa and Da Pozzo, 1961); towns' governments could participate to direct taxation by collecting additional land taxes. On average, the land tax amounted to around 6.7% of the yearly revenues from the land (Fossati, 1930), and was based on the Savoy land cadastre of the XVIII Century.⁹ Because of an arguably high fractionalisation of land property, the number of citizens subject to the land tax was far from small: evidence from two tax districts with detailed data on taxpayers shows that around 26% of citizens were subject to the land tax, so that the number of households hit by direct taxes could easily be much higher (Plebano, 1832, 1836). Importantly, the assessment of the tax base (and the correction of mistakes due e.g. to changes in changes of ownership) was not a task assigned to tax collectors, but to specific public officials of the central government entrusted with the task of verifying the accuracy of the tax base by keeping taxpayers' lists updated; such officials were in lower number compared to tax collector, and their districts did not overlap with tax districts.¹⁰

The tax collector was required to visit each town of the district at least once a month. However, transcripts from a debate of the Sardinian Parliament from 1854 (shown in Figure C.2) make clear that this rule was not respected: tax collectors' visits to their assigned towns were highly irregular or completely absent; as a consequence, tax payments were often taking place at the residence of the tax collector, forcing citizens to travel to the tax district capital in order to fulfil their duties towards the taxman.

Anecdotal evidence suggests the partition of the territory into districts was likely not implemented evenly and efficiently in the different areas of Piedmont. For instance, while officials of divisions like Saluzzo noticed how tax district capitals and their offices were

⁷Town-level taxation mainly involved consumption taxes and additional land taxes; while collected by the state's tax collector, their level was decided by the town's government, and they remained entirely at the town's disposal. Transfers from the central government were negligible throughout the XIX Century.

⁸At the time of the Italian unification in 1859, direct taxes in Piedmont amounted to 19% of tax revenues (Dincecco et al., 2011).

⁹While the French government invested substantial resources in the realization of a cadastre, by the time Piedmont re-gained its independence, the Napoleonic cadastre had been in less than a fifth of the towns of the Kingdom, so that the new government did not use it as an instrument to determine the tax base.

¹⁰Unfortunately, there are no comprehensive town-level data on the tax base; therefore, I implicitly assume it to be equal within the pairs of towns I use in my empirical design (described in section 4.2), conditional on controls. Reassuringly, within pairs, towns appear to be extremely similar in their population at several points in time, the number of high-income individuals allowed to vote in 1865, the height of drafted citizens, and their literacy rate, consistent with towns being equally rich despite being more or less treated.

located in the "most convenient" and central point of the district (Eandi, 1835), the excerpt in Figure C.3 tells how the allocation of towns to districts like the Baldichieri one (in the Province of Turin) looked less straightforward, as for some towns the capital was not the closest possible or the one with whom there were tightest commercial relationships, and more generally it was sometimes difficult to understand the unevenness in area and number of towns between the districts of the kingdom (Plebano, 1832).

The enforcement technology at the disposal of the central and the local governments in each town was the same, namely, the visits and the collection activity of the tax collector. In the absence of town-level data on taxes raised on behalf of the central government, taxes raised on behalf of local governments are still informative of governments' fiscal capacity before 1870. The technology under study can be clearly related to differences in tax enforcement, but may have also affected the decision of towns to raise taxes (or the tax rates applied): these two effects are both components of fiscal capacity, which I am not able to disentangle in this setting. Towns where tax enforcement was inadequate could have set higher tax rates to extract as much revenue as possible from the few taxpayers the collector succeeded in targeting; however, it is intuitive to think that towns with better tax collection took advantage of the situation to collect as much taxes as needed by the local government. In any case, even in the absence of differences in tax rates, differences in taxes raised could still emerge exactly because of differences in tax enforcement.

This tax collection system was kept in place in Piedmont from 1814 until the Italian unification, and on a temporary basis also in the first years following the unification process. Starting in 1870, collection of direct taxes was reorganised and homogenised across Italy: rights to collect taxes were assigned to private citizens on a town-level basis, hence making the districts irrelevant for tax collection purposes. Since 1861 the whole Italian territory was partitioned in *judicial* districts comparable to the districts used in the Sardinia Kingdom for judicial and tax purposes; this judicial administrative system was still in place in 1917.

2.2 Italian military draft in 1917

In 1917 the Italian Kingdom was participating in World War I for two years, fighting against the Austro-Hungarian Empire on its Northeastern border. The Italian male population was called for mass mobilisation, so that the vast majority of men able to serve in the Army were asked (compulsorily) to enlist.¹¹

Military service had been mandatory for every Italian man aged 20 (or older) since 1875 (L. 2532/1875), and for territories formerly under the Sardinia Kingdom since 1854 (L. 1676/1854). Before World War I, mandatory military service lasted two years. When drafted for military service, Italian men were asked to visit their division (or *circondario*¹²) capital

¹¹By the end of the war in 1918 around 5,9 million men were enlisted, out of around 12,2 million Italian men belonging to 27 birth cohorts that were called to serve (Ilari, 1990).

¹²A *circondario*, or a *division* as I call it in this paper, was the most relevant local administrative unit of the Sardinia Kingdom, with the ability to collect taxes and fund public goods (mostly roads). After the Italian unification, the

to undergo a medical examination; if they were considered suitable to serve, they were soon enlisted and sent to military camps.¹³ As a division included several judicial/former tax districts, for most citizens the district capital was not their division's capital.

Because of the considerable losses, in 1916 the Italian Army started to call men to serve before they turned 20. In particular, almost the entire cohort of 1899 was called to enlist before the age of 18.¹⁴

Evasion of the military draft was a widespread phenomenon. During the entire war, 11,31% of drafted Italian men (and 10% in the 1899 cohort) evaded the military draft (Ilari, 1990), by not showing up for medical examination or not enlisting after being considered suitable to serve. It was common for evaders to hide in mountain areas; however, many were found also in large cities like Turin (Melograni, 1969).

Draft evaders included the Italian emigrants who did not repatriate and did have a second citizenship, as Italian citizens were called to serve in the Army even when residing abroad. The Italian government organised and funded the repatriation of drafted migrant men and their families, so that around 300,000 Italian men returned from abroad to serve in the Italian Army, including more than 150,000 returning from North and Latin America (De Michelis, 1926).¹⁵

Before and during the war, drafted men could be exempted from military service for health reasons. However, in 1917 the Army considerably tightened its rules for granting exemptions, also calling previously exempted citizens to review the first visit and enlist them.

3 Data

In the following section I present the dataset I assembled and describe the main variables used in the empirical analysis. In Appendix B, I provide a detailed list of all the variables used in the analysis, their sources and their construction.

circondario remained present as an administrative unit but lost relevance, losing any power as a local government: essentially, it served to delimit judicial districts for the main courts of the Italian Kindgom, and hosted a in its capital a court and a representative of the central government.

¹³In times of peace, and during the first periods of the war, enlisted soldiers could be sent to every camp on the Italian territory; however, by 1917 the system was mixed and often soldiers would be enlisted in camps in their region, before being sent to the battlefront.

¹⁴Men born in the first four months of the year were called for medical examination in February 1917; men born in the following months were called in May 1917.

¹⁵While at the beginning of WWI it was not clear to many migrants what were their duties (Commissariato Generale dell'Emigrazione, 1924) even in the United States it soon became clear that Italian citizens were not exempt from the mobilization (Daly, 2021). Patterns of migrants were not unknown to the Army, which attempted to keep track of the residence of each citizen, also by using detailed data on the issuance of passports (a mandatory document to leave the country). In the whole sample of drafted citizens for the 1899 cohort, only 1.08% were reported to live abroad, and only 0.6% of them in Northern or Southern America.

3.1 Fiscal capacity

The main explanatory variable is the log of walking distance of a town from the capital of its tax district (the residence of its tax collector) during the 1814-1870 period, which I use as a proxy for historical fiscal capacity of the state in that town. I obtain the list of towns for the 1814-1870 period and their partition in tax districts from Stefani (1855).

I compute the walking distance (in meters) as the length of the least-cost path between the centroid of a town and the centroid of its (former) tax district capital. I follow Özak (2010) and determine the cost of moving through cells according to elevation, weather patterns and soil conditions,¹⁶ and using Tobler's hiking speed function to determine the walking-time cost of moving through cells.

As an alternative measure, I use the log euclidean distance of a town from its (former) tax district capital.

3.2 Draft evasion and other data from military records

Using the enlistment records from 1917 kept by the Turin State Archive (henceforth, TSA), I collect and digitise individual data on all the men called to serve in the Italian Army who belonged to the 1899 cohort and were registered as residents in the province of Turin.¹⁷ The 1899 cohort includes all men born in 1899 and a limited number of older men whose enrollment was deferred for medical reasons.

The men I observe in my sample were considered residents in the province of Turin if they were born there and never changed their residence, or if they were born out of the province of Turin but registered there as residents. The enlistment records also contain information on Italian emigrants born in the province of Turin who did not change their citizenship and were therefore asked to serve in time of war.

For each drafted man, the digitised data from enlistment records contain information on his personal details (name, surname, name of his father, height, literacy level), his town of birth and current town of residence, and information on his enlistment process (actual enlistment, exemptions for medical reasons, other types of exemptions, declaration of draft evasion). Using information from the enlistment records, a man is coded as a draft evader if he was declared an evader and such a declaration was never cancelled.¹⁸ Figure B.1 shows

¹⁶More specifically, Özak (2010) proposes a Human Mobility Index, computing the travel time (in days) keeping into account the aforementioned characteristics. In the main analysis of this paper I compute a path using the HMI procedure and use the log of length in meters of the path instead of travel time as main variable of interest. Results are robust to using the Human Mobility Index between a town and its former tax district's capital as an alternative measure.

¹⁷For the city of Turin, because of the large amount of data, I collect data on all the men born in another town, and a subsample of data for men born in Turin. As explained in section 4.3, men born in the city of Turin would be in any case excluded from the analysis sample.

¹⁸In many cases, drafted citizens were declared evaders when not attending their medical examination, but were acquitted if they showed up shortly later (e.g. after one month). In some cases, citizens were mistakenly classified as evaders if their change of residence had not been recorded, and were therefore enlisted in another town. In cases like these, men are classified as *non*-evaders.

an example of the original data from military lists.

Furthermore, I use data from enlistment records of all the men called to serve in the Italian Army that belonged to the 1899 cohort and were registered as residents in the province of Vicenza (digitised by the Vicenza State Archive, henceforth VSA). Data provided by VSA contain many pieces of information comparable to the data collected from TSA; however, they do not contain information on citizens' height and literacy, and provide less detailed information on their enlistment process.

3.3 Taxes collected by towns before and after 1870

I use data on all local taxes raised by tax collectors on behalf of towns belonging to the Sardinia Kingdom (Piedmont) before 1861. This sample's towns belonged to two Sardinia Kingdom divisions, Ivrea and Saluzzo. I collect these data from two different sources: for the division of Ivrea, I use archival records from 1847 kept by TSA on taxes collected in 1846; for the division of Saluzzo, I use data from Eandi (1835), which reports information on average tax revenues collected in Saluzzo between 1831 and 1834.

I also use data from Direzione Generale della Statistica (1887) and Direzione Generale della Statistica (1896) on town-level tax revenues after 1861 (in 1884 and 1895) for all the towns of the estimation sample.

3.4 Other variables

I measure turnout in 1865 local elections using information on eligible and actual voters reported by Statistica del Regno d'Italia (1867). I use information on actual voters and eligible voters from Corbetta and Piretti (2009), Direzione Generale della Statistica (1913) and newly collected data of the historical archive of the Italian Chamber of Deputies to build a town-level measure of turnout in 1913 national elections, as explained in Appendix G.1.

I also collect information on the population of Piedmontese towns in 1824 from Regno di Sardegna (1824), which I use as a proxy of population before treatment (pre-determined). From Santi (1902) I collect information, for each town, of its population in 1901 and the presence of police stations, post offices, and train stations. I also calculate towns' distances from Turin, their province capitals, their division capitals, the closest police station and international borders.

I use FAO-GAEZ (FAO, 2015) data to measure towns' land suitability for eight crops including wheat, rice, maize and potato. Finally, I obtain measures on towns' elevation, standard deviation in elevation, and land area from ISTAT. Additional town-level variables (used mainly in the analysis of section 7) and their sources are described in Appendix B.

4 Identification strategy

4.1 Empirical challenges

In order to estimate the effect of the historical fiscal capacity of the state on draft evasion in 1917, I test whether the distance of the town of birth of men from the historical residence (during the 1814-1870 period) of a tax collector predicts higher or lower likelihood that such men evade the military draft. Simple OLS estimates of this relationship pose two important empirical challenges to the identification of a causal relationship between fiscal capacity and draft evasion.

First, as towns hosting the residences of tax collectors were district capitals, they were likely to be chosen because of particular importance for the territory of their tax districts (e.g. because of higher population, higher economic importance, better connections to the road and railway network): if this were the case, towns closer to the historical residences of their tax collectors would also likely be larger, wealthier, and with different socio-economic structures. As a consequence, the distance of a town from the historical residence of a tax collector would likely be correlated with characteristics other than the historical fiscal capacity that could nonetheless affect the decision to enlist or to evade the military draft. Figure D.1 shows how, indeed, distance from former tax collectors' residences predicts significant differences in many town-level (even pre-determined) characteristics; in particular, towns farther from a former tax district capital tend to be significantly different from closer ones in their remoteness, and geography more generally.

Second, being the centre of an administrative division (albeit a very small one) the district capital was likely to host not just the residence of a tax collector, but also other relevant public offices since the time a tax collector was allocated there. If this were the case, the distance from a tax district capital would conflate both a fiscal capacity effect and effects from other dimensions of state presence (or state capacity).

In the following sections I describe how address these two issues.

4.2 Town-pairs fixed effects

A source of plausibly exogenous variation in historical fiscal capacity may come from discontinuities in distance from a historical tax collector's residence between towns close to each other, but belonging to two different (although adjacent) tax districts. Indeed, as districts were originally created according to thresholds on their population size and area, for towns on the border of tax districts the assignment to a district or the adjacent one might have been arbitrary, inducing quasi-exogenous variation in distance from a tax district capital between towns on the two sides of a border. Neighbouring towns on the two sides of a tax district border, being very close to each other, have similar characteristics, on average; however, because of their assignment to two different tax districts, they are likely to differ in the distances from where their tax collectors were living in the 1814-1870 period.

In order to maximise the comparability of towns that differ in their distance from a

former tax collector's residence, I compare *pairs* of neighbouring towns that straddle a tax district boundary,¹⁹ so that each town on the border of its tax district is matched with *every* other neighbouring town on the other side of its tax district.²⁰

Formally, when analysing aggregate town-level outcomes, I estimate equations of the following form:

$$Outcome_{tp} = \gamma_p + \beta \log Dist. Tax Collector_t + \mathbf{V}_t \delta + \varepsilon_{tp}, \tag{4.1}$$

where a town-level outcome for town t in pair p is regressed on $logDist. Tax Collector_t$, the distance of the town from the (former) residence of its tax collector, a full set of town-pairs fixed effects γ_p and (in some specifications) town-level controls \mathbf{V}_t .

When looking at individual-level outcomes, the estimating equation takes the following form:

$$Draft Evader_{itp} = \gamma_p + \beta \log Dist. Tax Collector_t + \mathbf{V}_t \delta + \varepsilon_{itp}, \tag{4.2}$$

where $Draft Evader_{itp}$ is a binary variable indicating whether citizen *i* born in town *t* in town-pair *p* evaded the military draft or not, γ_p is a full set of town-pairs fixed effects, $logDist.Tax Collector_t$ is the distance of the town of birth of a soldier from the former residence of its tax collector, and V_t are town-level controls for town of birth of a citizen.²¹

If a town borders with several other towns, it appears in the estimation sample multiple times: more precisely, a town appears in the sample as many times as the number of town-pairs it is in. Similarly, for the individual-level analysis, a drafted man appears in the estimation sample as many times as the number of pairs its town of birth is in. The presence of a single town within multiple pairs along a tax district border induces mechanical correlation in the residuals across town-pairs. Additionally, given the small size of the districts I look at, most towns (and as a consequence, in the individual-level analysis, most drafted citizens) appear on more than one border, inducing correlation in the residuals across several borders. To address this issue and avoid bias in the estimation of standard errors, I follow Dube et al. (2010) and Cantoni (2020) and throughout my empirical analysis I use two-way clustering by border and tax district.²²

Figure 2 illustrates the identification strategy using the border between the Corio and the Rivara tax district. The polygons bordered in white represent towns, and the blue line represents the border between the two tax districts. Centroids of district capitals are denoted

¹⁹I follow this approach as an alternative to comparing all towns on one side of a district border to all the towns on the other side of the border.

²⁰While in principle also pairs of towns within the same tax district could be used for this estimation strategy, towns on the border are those more likely to differ quasi-exogenously in distance from a tax collector's residence, and my focus allows me to maximise power by selecting pairs of towns for which difference in distance is sizeable.

²¹The focus on pairs of neighbourhood units on the border between administrative regions for identification and the use of individual-level outcomes is similar to Lichter et al. (2020).

 $^{^{22}}$ In some analysis the number of clusters in either dimension is below 50; following Lichter et al. (2020), in these cases I also present p-values estimated through the standard percentile-*t* Wild cluster bootstrap approach (Cameron et al., 2008).

by blue dots. The two towns of Rocca Canavese and Levone (with borders highlighted in red) are located on the border of their tax districts and they are adjacent to each other, so they enter the estimation sample and are matched in one town-pair (i.e. by sharing the same town-pair fixed effect). Because of geographical proximity, the two towns are likely to be extremely comparable in many characteristics;²³ similarly, also the men born in the two towns are likely to be similar. However, while Levone is 2.8 kilometres far from the historical residence of its tax collector, the corresponding distance for Rocca Canavese is almost twice as large. Equations 4.1 and 4.2 exploit variation in distances from historical residences of tax collectors only within pairs like the Rocca Canavese/Levone one, and test whether it predicts differences in likelihood to evade the military draft for men born in the pair.

4.3 Presence of other relevant offices

Being centres of administrative divisions, tax district capital could be hosting other relevant offices, on top of residences of tax collectors during the 1814-1870 period. In this case, the matching strategy described in section 4.2 would isolate the effect of all the offices hosted in the district capital, and not just the historical fiscal capacity effect.

In order to address these issues, I impose some restrictions on the towns I include in the sample. I start with all the Piedmontese towns of birth of drafted men I observe in my data, i.e. men born in 1899 who lived in the province of Turin in 1917.²⁴ First, I exclude tax district capitals from the estimation sample of my main analysis,²⁵ as their designation as capitals signals they might be too different from the other towns of the district (or the bordering district). Second, I exclude towns for which the distance from a former tax district capital coincides with distance from a division capital, which would be hosting other relevant public offices.²⁶ Third, to attenuate confounding effects from enforcement capacity of the state, I exclude towns for which the closest police station was located in the former tax district capital.²⁷

²³Including, for example, the proximity to the same relevant town: Rocca Canavese and Levone are less than 3 kilometres apart, and it is virtually indifferent to travel from Rocca Canavese to Corio or to Rivara (that are likely to be the relevant towns of the area, and to host amenities such as a train station or a local market).

²⁴I observe citizens born both in the province of Turin and in other Piedmontese provinces, who migrated to the Turin province afterwards. While for towns in the province of Turin I observe almost all citizens born in the town (as confirmed by out-of-the-envelope calculations), for towns in other provinces I observe only a few citizens. I do not exclude towns out of the Turin province so as to not reduce the sample size further; in any case, all the results are stronger when restricting the sample to Turin province only.

²⁵Because of the small size of the districts under analysis, tax district capitals were often on the borders of their tax districts.

²⁶I also include province fixed effects in every specification, so to focus on variation between towns differing only in their distance from a former tax collector's residence, and not only differences in the policies of their province. I choose fixed effects for provinces instead of divisions because of comparability between the Turin and the Vicenza samples (as better described in section 6.3) and because provinces were the relevant political unit, likely to enforce differential policies at the time of the war. Results using division fixed effects are generally stronger.

²⁷While this restriction does not rely on precise administrative division boundaries, I use closeness to a police station as the best approximation of the catchment area of the station, and impose this restriction to attenuate problems deriving by the overlap of a tax district and a police station catchment area, for which information are not available.

Finally, the allocation of tax collectors mapped one-to-one into the allocation of local judges. This implies that estimates of distance from a tax district capital could conflate both fiscal capacity and legal capacity effects. While it is impossible to disentangle the two effects in the context of Turin, in section 6.3 I estimate placebo regressions of the distance from judicial district capitals relying on the presence of judicial districts in the rest of Italy comparable to the districts in Piedmont, but that were never used for tax purposes.

5 Summary statistics and balance of observables

Table B.1 summarizes the characteristics of the towns and the individuals of the Turin sample, which I will use in the main analysis. Towns in my sample are arguably small (on average, with less than 1800 citizens in 1901), and only a minority of them hosted a police or a train station. The tax districts these towns belonged to during 1814-1870 were small, with around five towns per district; related to this observation, towns of birth of drafted citizens are on average less than 8 kilometres away from their former tax district capitals. Towns in the regression sample (selected as explained in section 4.3) tend to be less populated, and arguably more isolated (by looking at distance from province and division capitals, or at the presence of a train station) than the towns of birth of all the citizens drafted in 1917 in the province of Turin. Most of these differences are likely to be driven by the exclusion of tax district capitals from the regression sample: because of this, walking distance from a former tax collector's residence is almost twice as large in the regression sample than in the full sample.²⁸ Nonetheless, the share of draft evaders per town is comparable across samples.²⁹

Figure 3 presents estimates of β from equation 4.1 using (standardised) town characteristics as outcomes; Tables D.1 and D.2 report results from the same regressions. Under the identification assumption, these characteristics should be indistinguishable within couples of towns, on average; the estimates of such placebo regression should, therefore, be small and not significant. In each column of the tables I report a separate regression, where I always include town-pairs fixed effects (crucial for my identification strategy), province fixed effects, and I apply the sample restrictions described in section 4.3. Results show that controlling for town-pairs fixed effects succeeds in making towns comparable, despite their differences in distance from a former tax collector's residence: most pre-determined variables (such as geographical characteristics, distances from relevant cities, or population right-after the introduction of tax districts), and post-treatment variables (such as the presence of a train station, a post office or a police station) do not show significant differences between towns closer or farther from their tax collector in the 1814-1870 period; furthermore, most coefficients are small in size.³⁰

²⁸Where such a distance would be zero for district capitals.

 $^{^{29}4{,}52\%}$ in the regression sample, $3{,}94\%$ in the full sample.

³⁰Towns farther away from a former tax collector's residence are also significantly farther from a division capital, which was also the location were medical examinations for the draft took place. In all the results, distance from a division capital is included as a control in the regressions together with other geographical and pre-determined

Figure D.2 presents similar placebo regressions, using as outcomes the town-level averages of individual characteristics of drafted soldiers.³¹ The focus here is on characteristics which may directly influence the likelihood to show up (or not) for medical examination, or that may lead to differential ways of not enlisting. Within town-pairs, citizens born in towns farther from a former tax collector's residence do not differ significantly in their likelihood to be illiterate, or their height. In addition, within town-pairs, distance from a former tax district capital does not predict differences in two indicators of potential frauds in the process of the draft: the distance of interest does not predict higher share of citizens shorter than 150cm,³² or a smaller cohort size as a share of the population of the town.³³ In sum, the town-pairs fixed effects design greatly attenuates concerns that my measure for historical fiscal capacity could indeed be capturing the effects of other relevant drivers of the decision to enlist or evade the military draft.

6 Results

6.1 Distance from tax collectors and local taxation before 1870

As a preliminary result, Figure 4 presents evidence on the effect of log distance from a tax collector's residence on the amount of local taxes raised by tax collectors on behalf of towns' governments,³⁴ controlling for geographical and pre-determined characteristics.³⁵ in the 1814-1870 period, looking at towns from the provinces of Ivrea and Saluzzo.³⁶ Similarly to Heldring (2023) I use local tax revenues as a proxy for fiscal capacity at the local level.³⁷ Table E.1 presents results from the same specification. Because of the very small number

characteristics; when including it as the only other covariate in the baseline specification (on top of fixed effects) the coefficient of interest becomes larger in magnitude, and marginally significant. This suggests the effect of fiscal capacity on draft evasion is not confounded by its correlation with distance from a division capital.

³¹While these outcomes are available at the individual-level, they are absent for most draft evaders, as the more common way to evade the draft during the war was to simply not show up for medical examinations: as a consequence, more than 88 percent of the draft evaders would be missing when using available individual-level information. I use town-level averages of individual characteristics to alleviate concerns arising from nonrandom missing data.

³²This threshold was the minimal height possible to fight in the Italian Army, which had been lowered from 154cm in the first months of 1917. Anecdotal evidence suggests lowering the height was a common way to obtain exemptions for medical reasons before the war, however by 1917 exemption criteria had been substantially tightened.

³³Differences in such an outcome, assuming no differences in the age structure of towns, could indicate both errors in the way birth registries/draft lists were kept, or intentional tampering of these records.

 $^{^{34}}$ As described in section 2.1, towns relied on tax collectors to raise both direct and indirect taxes; according to calculations based on Fossati (1930) local taxes accounted for 15% of all the taxes a collector had to raise.

 $^{^{35}}$ Given the skewed distribution of local taxes and the presence of towns raising zero taxes, I use an inverse hyperbolic sine transformation of the outcome.

³⁶The relationship showed in the Figure 4 is based on the same specification defined in equation 4.1 and the sample restriction presented in section 4, plotting the binned scatter of the residual variation underlying the estimation of β after accounting for population and geographical controls.

³⁷Heldring (2023) uses tax revenues as an indicator of overall state capacity; analysing the German setting, its variable reflect the efficiency of local bureaucracies in and out of Prussia. In my setting, while being such taxes collected on behalf of towns, they were collected by the same collector as the central state: therefore, the measure I observe is linked to the fiscal capacity of both the local administration and the central state in the town.

of towns with tax data available, I first present results from all the towns in my sample not hosting a tax collector's residence (columns 1 and 2); I later show results for towns from a sample selected according to the criteria presented in section 4 (columns 3 and 4). Columns (1) and (3) of Table E.1 estimate equation 4.1 including only town-pair and province fixed effects, focusing on towns not hosting a tax collector's residence. In columns (2) and (4) I also include several pre-determined controls, including the population of the town in 1824, geographical characteristics of the town, and the distances from relevant towns (Turin, province capitals, and historical division capitals). In Figure E.1 and Table E.2 I present results from the same regressions, using per capita local tax revenues as outcome.

Results from Figure 4 and Table E.1 show a negative effect of distance from a tax collector on the taxes levied by towns, both in the extended and the restricted sample. In the extended sample, the estimate of the coefficient of interest is lower in size when including all the controls, but remains significant at the 5% level. Interpreting inverse hyperbolic sine transformations as percent changes, a 1% increase in distance from tax collector's residence leads to 0.44%-0.69% reduction in taxes levied by municipalities; a 1km increase in such a distance leads to an around 9%-15% decrease in local taxes raised. When restricting the analysis to the sample defined in section 4, the estimated coefficient is not stable across specifications, and it is not significant in the baseline specification; however, it is negative and significant (on top of being substantially large in size) when controlling for geographical and pre-determined characteristics; in such a specification, a 1km increase in distance from a former tax collector's residence leads to an around 30% reduction in taxes levied. Elasticities are similar when looking at per capita taxation. The substantially small sample size (in particular if compared to the number of fixed effects included in the regression) makes results from such analysis only suggestive, but still consistent with the government having higher fiscal capacity in towns closer to the tax collector's residence.

As noted already in section 2, the measure I observe (local tax revenues) represents the final outcome of the taxation activity of towns' governments, resulting from many components of fiscal capacity, from the effectiveness and the efficiency of tax enforcement to the amount of taxes a town decided to levy in the years under analysis. Yet, while potentially arising from many channels and being estimated on a very low number of towns and clusters,³⁸ evidence from Table E.1 suggests how distance from a tax collector was indeed associated with citizens paying fewer taxes in the 1814-1870 period, and supports the strategy of using distance from the residence of a tax collector's residence as a measure of the government's fiscal capacity in the 1814-1870 period.

6.2 Historical fiscal capacity and draft evasion in WWI

In Table 1 I present estimates of equation 4.1 testing the effects of distance from former residences of tax collectors on the share of draft evaders from the 1899 cohort. Column (1)

³⁸The two regression samples include 120 or 36 towns, respectively, located in 36 or 16 tax districts.

of Table 1 shows results including only town-pair and province fixed effects. In column (2) I include controls for town-level geographical characteristics (which may have influenced the location of tax collectors' residences before 1870), while in columns (3) I include controls for town-level averages of individual-level variables (height, literacy, being exempted for health reasons, resident abroad), and several town-level variables potentially determined after the implementation of the treatment (e.g. presence of other public offices; presence of a train station; population in 1901), which may nonetheless affect the decision to evade the military draft.³⁹ Results show that towns farther from the former residence of a tax collector have significantly higher shares of draft evaders born there in 1899. The estimated effect of distance almost doubles in size when including town-level geographical controls, and it is significant (on top of being larger in size) also when including controls for individual characteristics and post-treatment town-level characteristics. Depending on the specification, a 1% increase in distance from a former tax collector's residence leads to a 0.015-0.036 p.p. increase in the share of evaders in the town; the effect is non-negligible in size, as a 1km increase in distance at it s mean corresponds to an around 0.36 p.p. higher draft evasion rate when controlling for geographical and pre-determined controls, a 7.5% increase compared to mean evasion in the estimation sample.

In Table 2 I exploit the individual nature of my draft evasion data, and report estimates of equation 4.2, testing the effects of distance from former residences of tax collectors on the likelihood of individual citizens to evade the military draft, as measured by a binary individual-level variable. Compared to the town-level analysis such an exercise is also useful because estimating effects on a binary outcome variable allows isolating the extensive margin from the intensive margin.⁴⁰ In order to inspect how the effect of historical fiscal capacity compares to (or is correlated with) the capacity of the government to enforce the military draft in 1917, I also augment the specification in equation 4.2 with the inclusion of the distance of the town of residence of a citizen to the closest police station. Column (1) estimates the association between the i.h.s. transformation of distance from a former tax collector's residence⁴¹ and the likelihood of draft evasion without covariates, column (2) does the same for distance from the closest police station, and columns (3) to (5) report the two coefficients together, adding covariates in the same fashion as in Table 1. Consistently with results at the aggregate level, Table 2 shows a negative and significant effect of my proxy of historical fiscal capacity on the likelihood that a citizen born in the towns evades the military draft. Additionally, the estimates show that the historical fiscal capacity effect

³⁹Figure 5 plots the binned scatter of the residual variation underlying the estimation of β in the specification from column (1) of Table 1.

 $^{^{40}}$ In section 6.4 I also present town-level results with a binary indicator of draft evasion as an outcome.

⁴¹In this table I include the i.h.s. transformation of distance from a police station as my research design does not require to exclude from the sample towns of residence hosting a police station (which would also reduce sample size substantially, as towns of residence are in general larger than towns of birth); for comparability, I therefore use the i.h.s. transformation of distance from former tax collector's residence. As tax district capitals are still excluded from the sample, results are essentially unchanged compared to a specification with the logarithm of such distance.

remains positive and significant also when controlling for distance from a police station; remarkably, the estimated size of the effect is essentially unchanged with respect to the specification without distance from a police station.⁴²

As expected, distance from a police station is positively (and significantly) associated with the likelihood of evasion of the military draft; however, its estimated effect is substantially smaller compared with the historical fiscal capacity effect. It is important to notice that police stations are not allocated randomly on the territory: as a consequence, in the absence of information on their precise catchment areas, the effect of distance from them cannot be estimated causally in this context. Furthermore, while proximity to a police station may look like an intuitive proxy of law and draft enforcement (also, for draft enforcement, anecdotal evidence is strongly consistent with the territorial police having an important role in chasing evaders), it is likely that in times of war the allocation of policemen on the territory might be altered, so introducing measurement bias in the enforcement capacity of the Italian state. Nonetheless, results from Table 2 show how the effect of historical fiscal capacity is distinct from effect of differential capacity to enforce the draft.⁴³

My findings indicate that historical fiscal capacity is associated with sizeable effects on compliance with the military draft in the long run. To the best of my knowledge, this is the first paper relating fiscal capacity to military draft; furthermore, it does so focusing on a previously unexplored setting, i.e. Piedmont and institutions of the Sardinia Kingdom. However, the effects I estimate are comparable to (and slightly smaller than) results from previous research that analysed either draft evasion as an outcome, the effects of taxation on citizens' attitudes towards the state, or the cultural consequences of exposure to high state capacity. In Appendix E.1 I explicitly compare my result to such research, finding similarities with studies from both the Italian, European and Sub-Saharan African settings.

6.3 Placebo test: legal capacity and draft evasion in WWI

In this section I replicate the aggregate analysis proposed insofar on another sample, the one of citizens drafted in the Italian province of Vicenza (the Vicenza sample),⁴⁴ applying sample restrictions similar to the ones adopted for the main sample and described in section 4.3.⁴⁵ As mentioned in section 2.1, citizens born in the province of Vicenza (like every citizen

⁴²The estimated coefficients of the effect of my proxy for historical fiscal capacity without controlling for distance from a police station (not shown here) are remarkably similar to the ones shown in columns (4) and (5) of Table 2 also for specifications with covariates.

 $^{^{43}}$ Note that the estimation sample does not include individuals born in towns that had the closest police station in the tax district capital.

⁴⁴Table F.2 presents summary statistics of town-level characteristics in the Vicenza sample.

⁴⁵Provinces of Veneto like Vicenza, from its annexation to Italy until World War I, had a slightly different administrative structure than other Italian provinces: they were partitioned in *distretti*, and not divisions (*circondari*). As *distretti* were much smaller than *circondari*, excluding towns for which a judicial district capital hosted a *distretto* would reduce the number of observations too much. As a second-best alternative, for the Vicenza sample, I exclude towns for which the judicial district capital was also the province capital. Results from the Turin sample using this alternative sample restriction are consistent with the ones shown in section 6.2.

living in an Italian territory not belonging to the Sardinia Kingdom before 1861) were also living in towns partitioned in districts; however, these districts were only used for judicial purposes, with the district capital hosting a court in charge of handling minor cases. As a consequence, this analysis provides insights on how some components of *legal* capacity (both historical and contemporaneous, as proxied by distance from a local court) may have on the likelihood of citizens to evade the military draft, as well as the effect of distance from a relatively large town (i.e. relevant enough to host a minor court).

In Table 3 I report estimates of equation 4.1 on the Vicenza sample,⁴⁶ while Figure 6 plots the relationship between distance from a judicial district capital and the share of draft evaders in a town of the Vicenza sample (controlling for town-pair FE and pre-determined characteristics). Results from the Vicenza sample suggest the effect of distance from a (judicial) district capital is very different from the one estimated on the sample from Turin (which potentially conflates legal capacity and historical fiscal capacity). Coefficient estimates are mostly negative,⁴⁷ smaller in size compared to the Turin ones, and consistently not significant.⁴⁸

The estimation sample from the province of Vicenza has around 14% fewer draft evaders than the province of Turin, which could be a result of the potentially different coding of the *Draft Evader* variable performed by VSA. It must also e noted that Vicenza was very close to the Italian battlefront during WWI, hosting the headquarters of many regiments; overall, the military draft may have been more enforced differently in Vicenza compared to the Turin province. However, despite a slightly lower baseline evasion level, equations 4.1 and 4.2 allow to recover the causal impact of legal capacity on draft evasion provided that (within town-pairs) distance from a local court is not correlated with differential enforcement of the draft, a plausible assumption given that most districts were far from the battlefront and local courts were not responsible for handling cases of draft evasion or desertion. While the specificity of the province under analysis suggests some caution in interpreting results from it, evidence from the Vicenza sample strongly supports the claim that the effects observed in Turin should be attributed to historical fiscal capacity, and not to proximity to local courts.

It may be surprising to see a variable correlated to legal capacity having no effects on draft evasion, a decision linked to rule-following behaviour. However, the particular nature of the courts under analysis can explain the effects found in the Vicenza sample. Indeed, local courts in the judicial district capitals only handled minor criminal and civil cases, while most cases (and the relevant ones) were handled by courts located in the division's capitals; as a consequence, it is plausible that the legal capacity of the Italian state (and Piedmontese

⁴⁶The set of covariates used as control variables in this analysis is slightly smaller than the one used in the analysis for the sample of Turin, as I miss two individual-level controls for every drafted citizens, i.e. height and literacy level. I also use imputed population in 1843-1844 (computed as explained in section B.3) in place of population in 1824, a piece of information available only in the Turin sample.

⁴⁷The estimated coefficient is positive only in column (3), where I control for post-treatment town-level characteristics.

⁴⁸ in Table F.1 I present results from individual-level regressions on the Vicenza sample, that confirm how the estimated effect of legal capacity does not seem to affects draft evasion, or at best has a very different effect on it.

state before) could not be proxied by the distance from a local court. As a consequence, the analysis from the Vicenza sample cannot provide conclusive evidence on the effect of legal capacity on culture; however, it is valid to attenuate concerns about legal capacity driving the effects found in the Turin sample.

6.4 Robustness

As an alternative empirical strategy, I follow Cantoni (2020) and estimate the effect of historical fiscal capacity on draft evasion relying on a less demanding specification. In particular, I rely on a border fixed effects design where I regress the share of draft evaders born in a town on the distance of the town from the former residence of its tax collector and a full set of border fixed effects, equal to one for all the towns on the two sides of the boundary between two tax districts.⁴⁹ As shown in Table H.1 results from the border fixed effects analysis are consistent with the ones obtained using the town-pairs fixed effects design: estimated coefficients are positive and significant once controlling for geographical and pre-determined characteristics. While the town-pairs design is useful in maximizing the comparability between towns differing in distance from a district capital, using border fixed effects attenuates potential concerns arising from the presence of multiple observations per town (1,52 on average compared to 1,72 in the main analysis).

I also provide evidence that the results from the Turin sample presented in this section are qualitatively robust to alternative measures of historical fiscal capacity, of draft evasion, or to other sample selections.

In columns (1) and (2) of Table H.2 I present results using the log of euclidean distance from a former tax collector's residence as main explanatory variable, while in columns (3) and (4) I use log cost-distance (computed using the Human Mobility Index by Özak, 2010); finally, in columns (5) and (6) of the same table I show results where I measure historical fiscal capacity with a binary variable, indicating whether a town (within a couple) is farther from its former tax collector's residence compared to the other town. The effect of historical fiscal capacity on draft evasion remains negative regardless of the measure adopted; using a binary treatment, I disregard important variation in the main distance of interest so that coefficients of distance from tax collector's residence are imprecisely estimated; nevertheless, they are consistently positive and non-negligible in size.

My research design leads me to focus on small towns: as a consequence, the number of men drafted in 1899 in each town is arguably small (17 on average), and so is the number of draft evaders (1 on average). Therefore, my main outcome of interest (the share of draft

Share
$$Draft Evaders_{tb} = \gamma_b + \beta \log Dist. Tax Collector_t + \mathbf{V_t}\delta + \varepsilon_{tb},$$
 (6.1)

 $^{^{49}\}mathrm{For}$ this analysis, I estimate the equation:

where b denotes a border between two districts the town is on. Note that, given the small size of the districts, several towns share their boundaries with several borders between districts, so the standard errors are again clustered both at the tax district and district-border level.

evaders born in a town) reflects differences in draft evasion both on the extensive and the intensive margin. While the individual-level analysis shown in Table 2 provides results fully consistent with the ones from the aggregate-level analysis, in Table H.3 I show that, even if the discretisation of the outcome makes estimates less precise in some specifications,⁵⁰ historical fiscal capacity has a negative effect on draft evasion in a town also when focusing on the extensive margin (namely, using as an outcome a binary variable for the presence of any draft evader born in a town). Results are essentially unchanged compared to Table 1 when using the inverse hyperbolic sine transformation of draft evasion as outcome variable.

To further alleviate concerns related to multiple counting of towns in my matching procedure, in columns (1) and (2) of Table H.4 I replicate the main analysis but use only one match for each town in the regression sample. More specifically, I use the following procedure: first, I select towns and town-pairs using the same criteria as in section 4.3; second, I match them to all the contiguous towns on the other side of the district border (that respect the criteria described in section 4.3); third, for each town, I keep only the pair with the closest⁵¹ matched town on the other side of the border.⁵² Results are remarkably comparable in magnitude to the ones obtained using all the pairs, while being less precisely estimated; therefore, estimates from such analysis provide additional evidence that the results of the main analysis are not simply driven by the presence in the sample of multiple observations per town.

In columns (3) and (4) of Table H.4 I slightly vary the sample composition, excluding towns whose tax collectors' residence coincided with the province capital (instead of the division capital); results are similar with the ones presented in section 6.2. Consistent results are also shown in columns (5) and (6) of Table H.4, where I use more demanding division fixed effects. In columns (7) and (8) of the same table I also allow for tax districts' capitals to be in the sample, provided they did not host a police station. Using this (unbalanced) sample, the estimated effect of distance from a tax collector remains positive, despite being imprecise and of substantially smaller size compared to the main results.

In columns (9) and (10) of Table H.4 I find consistent results when restricting my analysis to towns of birth of citizens in the province of Turin, therefore excluding all the towns of Piedmont out of this province where migrant men living in Turin were born. Such analysis addresses the concern that aggregate analyses based on the share of draft evaders out of the total number of draftees in my sample would not be reliable in towns for which the total number of draftees cannot be determined in a precise way. By focusing on towns of birth in the province of Turin, while reducing the sample size, it can be assured that the

⁵⁰Significance of such analysis is higher when restricting the sample to towns of birth of the province of Turin (reaching 10% and 5% in the specifications without controls and with geographic controls respectively).

 $^{^{51}}$ I measure the closeness between two towns using the euclidean distance between their centroids.

⁵²Note that, while reducing the number of observations, following this approach some towns are still repeated multiple times in the dataset: this is a consequence of a town being the closest match for more than one other town. Therefore, I adopt the same empirical specifications described in 4.2, including the double clustering procedure.

overwhelming majority of men of the 1899 cohort will be included in the sample. Such results reassure that the relevant variation in the results comes from the vast majority of observations for which measurement quality of the outcome variable should not be a concern.

Table H.5 shows that my main results are robust when accounting for spatial correlation using several cut-off distances. Finally, Figure H.1 provides estimates of Oster (2019)'s degree of selection on unobservables relative to observables, which provides evidence that my results are unlikely to be driven by unobservable heterogeneity in town's characteristics.

7 Mechanisms

7.1 Channels of transmission: culture vs. institutions

In this section, I provide evidence on the potential channels of transmission of the effect of historical fiscal capacity in Piedmont. On the one side, while institutions for tax collection under analysis were no longer in place in 1917, they might have shaped the subsequent institutions of towns of the Italian Kingdom, including institutions potentially affecting the decision to enlist in the Army, or evade the draft (e.g. draft enforcement or rule enforcement in general). On the other side, historical fiscal capacity may affect military draft evasion in the long run because it affects the culture of individuals, and potentially of a community in a persistent way (see e.g. Nunn and Wantchekon, 2011). I test for culture as a channel of transmission focusing on the sample of internal migrants in my data. Such citizens were born in a given town, but were living in a different town at age 18. As a consequence, studying their decision to evade the military draft allows to disentangle institutional factors (which pertain to the town of residence of the citizen) from cultural determinants of the decision to evade the military draft (more likely to be a characteristic of the town of birth of citizens, and arguably of their parents). I therefore estimate equation 4.2 on the sample of migrants only, and include town-of-residence fixed effects in the regression. In such a demanding specification, I estimate the effect of historical fiscal capacity on draft evasion leveraging only variation between individuals born in two neighbouring towns (on borders of their tax districts), but living in 1917 in the same town. Town-of-residence fixed effects allow to estimate the cultural component in historical fiscal capacity, as every external (institutional) factor would be the same for citizens under analysis, while historical fiscal capacity in their towns of birth would be different.

In Table 4 I present results from such analysis. While coefficients are less stable than in the main results (possibly because of the smaller sample size), they are all substantially larger than those estimated on the sample of natives and migrants; furthermore, they are statistically significant when controlling for geographical and predetermined town characteristics. These results are consistent with the hypothesis that the effect of fiscal capacity persists in time by affecting the culture of communities exposed to it, rather than through persistent differences in institutions.

A way to test directly for institutions as a channel of transmission is to study whether

historical fiscal capacity had a persistent effect on the ability of towns' governments to collect taxes in the long run, despite the disappearance of the tax districts under analysis.⁵³ Indeed, towns may have inherited differences in fiscal capacity from the pre-1870 system, with towns previously closer to the tax district capital more able to retain best practices in tax collection. Using data on town-level taxation in 1884 and 1895 I investigate whether transmission through institutions may be at play, and whether distance from a tax collector's residence before 1870 predicts differences in taxes collected even when tax districts did not exist anymore; importantly, being such data available for every Italian town, such an effect can also be tested in the sample of towns used in the main analysis having draft evasion as the outcome of interest. In particular, I estimate the equation:

$$Local Taxes_{tcy} = \gamma_{cp} + \beta_1 \mathbb{1}_y^{1846} Distance_t + \beta_2 \mathbb{1}_y^{1884} Distance_t + \beta_3 \mathbb{1}_y^{1895} Distance_t + \lambda_y + \sum_{p \in P} \mathbf{V_t} \mathbb{1}_y^p \delta_p + \varepsilon_{tcy},$$

$$(7.1)$$

where t indicates the town, c a town-couple (or town-pair), $p \in P$ a period, and y a year; λ_y are year fixed effects, P can be two periods (pre- and post-1870) or tax years (1846, 1884 and 1895) and I allow for the effect of control variables V_t and pair fixed effects γ_{cp} to vary before and after the removal of tax districts.⁵⁴ β_1 , β_2 and β_3 estimate the separate effects of log distance from a tax collector in the 1814-1870 period on taxes collected in 1831/1846, 1884 and 1895, respectively. Results from Figure 7, where I plot the coefficients from equation 7.1 both for the Ivrea and Saluzzo sample and for the main estimation sample, show that even if in both cases the distance from a former tax collector's residence still has a negative effect on taxes collected by towns, the estimated coefficients in 1884 and 1895 are substantially smaller in size compared to the ones estimated before 1870. This suggests a dissipation in time of the specific ability of town governments to collect taxes as a result of proximity to tax collectors. On top of direct evidence on persistence through fiscal capacity, additional indirect evidence is not consistent with persistence through institutions: indeed, because of the sample selection and the results in section 5, the effect is unlikely to be driven by fear of police enforcement that transmits over time, and such an hypothesis is also supported by the results from the placebo test on the Vicenza sample, which shows no effect of distance from a minor court on draft evasion.

Taken together, results on migrants' likelihood to evade the draft and on the evolution of local taxation strongly suggest that historical exposure to higher fiscal capacity affects draft

⁵³Another strategy would involve studying how the distance of the town of residence of a drafted man from a former tax collector's residence affects the decision to evade the military draft. However, most migrants settled in larger towns and cities like Turin, which often were former tax district capitals: in these cases, distance from a former tax collector's residence would be zero, and as a consequence, there is too small variation in such distance to estimate its effects on migrants' decisions credibly.

⁵⁴Because of the small number of observations in the Ivrea and Saluzzo sample, in my preferred specification I present results interacting control variables with pre- and post-1870 fixed effect; as shown in Figure E.2 and Table E.3, patterns are similar also when interacting them with years fixed effects.

evasion mainly through a change in the culture of affected communities.

7.2 Perceived state capacity and returns to participation

Once ascertained that culture is likely to be the prevalent way through which the effect of fiscal capacity was transmitted in time, I explore which phenomena may drive the positive relationship between fiscal capacity and draft evasion in World War I.

Enlistment in the Army, in a setting as a war between national states, can be intuitively seen by citizens as a way to contribute to the defence of a common good, namely their state and, ultimately, its independence. On the other hand, being fiscal capacity the ability of the state to collect taxes effectively, states with higher fiscal capacity are also likely to send a signal of higher state capacity overall: indeed, a government able to enforce tax collection effectively can be perceived as having the ability to implement effectively also other public policies. If fiscal capacity makes citizens update positively their perception of the state: an action they could concretely participate in through military enlistment. It follows that historical fiscal capacity can induce lower draft evasion in the long run through a persistent improvement in citizens' perception of state capacity and an increase in the perceived returns to participation in war.

In the absence of data on citizens' perception of state capacity or governments' performance, a way to test for such a mechanism passes through citizens' perceived returns to participation in the state's activities and, therefore, willingness to be involved in the political process. Indeed, a citizen perceiving the state as worth defending may be also motivated to express her voice in the state's activities, as high state capacity makes it more likely that the government can do something meaningful to accommodate the citizen's needs.

Given the setting, I use newly digitised data to test for the effect of historical fiscal capacity on political participation through voter turnout in two elections, namely elections for the town's council in 1865 (local elections) and for the national Parliament in 1913 (national elections).⁵⁵

The tax collector of the Sardinia Kingdom was encharged to collect taxes on behalf of both the central and the local government; as a consequence, were fiscal capacity connected to political participation, the effect on turnout should be detected not only in national elections but also in local ones. Looking at local turnout also allows to test the mechanism under analysis at a point remarkably early in time, namely in 1865, when the tax collection system under analysis was still in place; additionally, the local electoral franchise in Italy was larger than the national one (which was restricted to elites, as it was common in the XIX Century): in towns of my sample, around 26% of the male population had the right to vote for the city council.

Voter turnout in national elections is a measure more directly connected to the rela-

⁵⁵Data sources and variables' constructions are described in sections 3 and G.1.

tionship between citizens and the central state: thereby, it is more intuitively related to enlistment in the state's Army. As the research design leads to focus on small towns, 1913 is the only year before WWI for which (because of the 1912 extension of the franchise) there are enough observations at the town level to make it feasible to test for this mechanism.⁵⁶ An analysis based on 1913 electoral data is interesting because it allows studying whether a relationship between fiscal capacity and political participation exists in the long run, and more precisely exactly before the beginning of the Great War.

In Table 5 I present results on local and national turnout, both without controls (columns 1 and 3) and controlling for geographical and pre-determined characteristics (columns 2 and 4). In both elections, being farther from a tax collector's residence before 1870 predicts lower voter turnout; the effect is significant (marginally in 1913) and robust to the inclusion of controls. While the franchise (especially in 1865) depended on literacy and income, Table G.1 shows that historical fiscal capacity does not drive differences in the share of citizens allowed to vote. The positive effect of fiscal capacity on political participation (both in the short and the long run) supports the interpretation of fiscal capacity as a driver of higher perceived state capacity, which induces citizens to participate more both in the political life of institutions and to the effort required to defend them.

While, reassuringly, I find consistent results between the outcomes of my analysis, the two indicators of turnout differ both in the type of elections they refer to, the franchise, and the year elections took place; all these channels can explain the differences between the effects estimated when looking at local or national turnout.

Previous studies related voter turnout to several different phenomena (see e.g. Rei, 2020; Durante et al., 2023; Marciante, 2023); yet, evidence from this paper is mostly consistent with results from Weigel (2020), which finds that the introduction of tax collection in Kananga (DRC) spurred political participation, and updating beliefs about state capacity as a mechanism at play. Evidence from my analysis shows how this effect may persist in time, and end up affecting other margins of interaction between citizens and the state.

7.3 Alternative mechanisms

A plausible interpretation of the main result links the decision to go to war to the provision of public goods, or public funds more generally (Caprettini and Voth, 2023). Indeed, if higher fiscal capacity in a town allowed to fund more public goods, such provision could induce a sense of gratitude towards the state; this may in turn explain lower levels of draft evasion in towns closer to a former tax collector's residence. In Table 6 I test for this channel, looking at the effect of historical fiscal capacity on several indicators of public goods provision that may

⁵⁶Both the electoral laws of 1882 (in place until the 1909 elections) and the one of 1912 (in place in 1913) required towns with less than 100 *voters* to be aggregated in a unique polling station with other towns; with the extension of the franchise, more and more towns gained an independent polling station. Nonetheless, having towns in Northwestern Italy a relatively low number of citizens and voters, because of the criteria of the 1882 law, more than 35% of towns in my sample did not have an independent polling station. As votes were only recorded at the polling station-level (or precinct): this makes it unfeasible to perform the empirical analysis on electoral data preceding 1913.

have been particularly salient to citizens, both under the Piedmontese state and the Italian one (before and after 1861). In columns (1) and (2) I look at funding received by charity institutions and hospitals (measured in 1840) that were funded by the central government. while in columns (3) to (6) I study effects on total spending on primary schools in 1863 (when funds for such schools came almost entirely from town governments) and on kindergartens in 1869; in columns (7) and (8) I look at the presence of sewage infrastructure in the town, and in columns (9) and (10) I test for effects on the number of physicians paid by the town's government, using data from Direzione Generale della Statistica (1886).⁵⁷ Overall, looking at measures from both before and after the Italian unification, the results show that towns subject to higher fiscal capacity were not receiving more public goods provision at the central or the local level. Also, this evidence complements the findings in columns (5) and (6) of Table D.2, where I show that towns where historical fiscal capacity was higher, were not significantly more likely to have a post office or a train station, two other outcomes that can be clearly related to public good provision. Taken together, evidence from these indicators do not support the hypothesis that differences in the provision of specific public goods between towns drive the effect of historical fiscal capacity on draft evasion. This result seems difficult to reconcile with observed differences in local tax revenues, which can be interpreted as overall local public spending. However, even in a setting where governments were not extractive and public investments existed, it is likely that public spending was mainly targeted to mandatory expenses (such as primary schools in XIX Century Italy), to goods that were provided at a level higher than the town (for instance, bridges between towns, provincial roads, or the national railroad, or national defense through the Army's activity), or even partitioned into a large number of different spending chapters (which may differ between towns, when looking at local governments' spending). All these phenomena can explain why differences in public good provision are difficult to detect within pairs of towns.

A different, cultural driver through which historical fiscal capacity may affect the decision to enlist for war is higher patriotism of citizens exposed to more effective tax collection. The rationale for this mechanism is the following: provided that tax collection sends a signal of higher state capacity and citizens perceive higher returns from participating in war (consistent with the evidence presented in section 7.2), military enlistment may be a consequence of higher patriotism if state presence (signalled by the presence of tax collectors) induces citizens to *identify* more with the central state. In columns (1) and (2) of Table 7 I test for the effect of historical fiscal capacity on patriotism using as outcomes the share of citizens born in a town who received a Medals of Honor during WWI.⁵⁸ Receiving a Medal

⁵⁷The role of investment on public schools on draft evasion may be twofold: first, it was a clear sign of public good provision from the government; second, as shown by several pieces of evidence, it may have been conducive in instilling a patriotic attitude in pupils and a building a unique national identity (Chilosi, 2007; Alesina et al., 2021; Blanc and Kubo, 2021).

⁵⁸Data on Medals of Honor pertain only soldiers who died in war; the construction of the variable is explained in detail in Appendix B.3.

of Honor is an outcome that signals heroic actions, and therefore more correlated to genuine patriotism than compliance with the social contract. While the baseline estimate shows that distance from a tax collector has a negative and marginally significant effect on the share of medals received by citizens, the coefficient is unstable and turns positive (while not significant) when controlling for geographical and pre-determined characteristics. I further investigate for higher patriotism in towns with higher historical fiscal capacity looking at first names of drafted citizens, and computing the share of drafted citizens bearing the name of the King or former kings of the Savoy family.⁵⁹. As shown in columns (3) and (4) of Table 7, distance from the former tax collector's residence does not predict sizeable or significant differences in the share of citizens named after a Savoy king. Overall, these results do not support the hypothesis that in the Piedmontese setting historical fiscal capacity of the state increased the patriotism of citizens, nor that it affects the enlistment or evasion decision through this channel.

My results would be driven by an alternative mechanism if higher fiscal capacity led to higher fairness in taxation, which may have induced citizens to positively update their attitudes towards the state, or even increase their trust. While systematic tax collection could have been perceived as fairer in towns where fiscal capacity was higher, tax collectors in Piedmont were only asked to collect taxes, and were not involved in broadening the tax base or correcting inaccuracies of the cadastre: their role was to enforce tax obligations, and not increase the fairness of the tax system. As a consequence, such a mechanism is unlikely to be at play given the specifics of the Piedmontese tax system.

Finally, the capacity to collect taxes may affect draft evasion in the long run through the emergence of rule-following norms that prescribe how to interact with the central state, and potentially to obey its requests. Citizens exposed for more than 50 years to higher fiscal capacity probably got accustomed to paying taxes and recognising the legitimacy of the state's financial requests: such behaviour would generate a norm of tax compliance. If respecting norms of tax compliance produced spillovers on other norms of rule-following (in a logic similar to Keizer et al., 2008), then fiscal capacity would be able to foster the emergence of a general norm of rule-following, capable of affecting also the decision to evade the military draft; evidence from a similar mechanism has been found by Heldring (2021). While such a mechanism is also consistent with my findings, no town-level data from the XIX Century allows me to empirically confirm it or disprove it; nonetheless, the emergence of norms of rule-following (or even obedience) and the updating mechanism described in section 7.2 are not mutually exclusive.

⁵⁹First names have been widely used to measure cultural traits: indeed, naming of a child is an important decision for parents, who can in this way signal their cultural preferences (Fryer Jr and Levitt, 2004; Fouka, 2019; Abramitzky et al., 2020; Russo, 2019; Bazzi et al., 2020; Beck Knudsen, 2019; Varnum and Kitayama, 2011; Giuliano and Tabellini, 2023). The use of names of Savoy kings in this setting is convenient as they were relatively few and infrequent. In addition, compared to the share of Medal receivers, this measure pertains to all drafted citizens, and not only dead male citizens who did not evade the military draft.

8 Conclusion

In this paper, I provide evidence that higher fiscal capacity of the state leads citizens to higher compliance with rules that ask them to perform costly actions. This increase in compliance is visible generations after the removal of the institutions that created differences in the capacity of the state to collect taxes. By enforcing taxes, the state can induce compliance with the law also in domains that are not directly related to taxation.

Crucially, tax collection improves compliance by changing the way citizens interact with the state, promoting their willingness to participate both in the political activity of the state and to other activities such as the state's defence.

Through higher tax revenues, a state with high fiscal capacity is clearly able to build higher capacity to enforce laws and implement public policies effectively. However, evidence from this paper shows that even in cases where fiscal capacity does not generate differences in other dimensions of state capacity (such as within pairs of neighbouring towns), tax collection may prove to be a precious tool for the government by improving citizens attitudes towards the state and, eventually, fostering the emergence of compliance.

While my analysis is based on a sample of towns in Northwestern Italy, my results are comparable in sign and magnitude to research that studied the interaction between institutions and culture in several other settings and periods. At the same time, the focus on Piedmont allows to isolate the effect of fiscal capacity in an observational setting, and look at its consequences in the long run.

My results point to a novel interpretation of fiscal capacity's contribution to the state capacity building process: indeed, through increased compliance with the law, fiscal capacity can indirectly facilitate state building by reducing the resources that a state needs to allocate for law enforcement. Furthermore, looking at compliance with the military draft, this paper shows that fiscal capacity contributes to state capacity also by improving military enlistment (Qian and Tabellini, 2021), a component of state capacity itself, as it makes a state more able to fight –and defend itself– during wars.

It is still to be understood how other components of state capacity (as, for instance, legal capacity) affect cultural norms, and which component of state capacity has the larger effect on culture. Furthermore, this paper leaves open the question of how long run effects of fiscal capacity may influence cultural norms more generally, and not only in situations when the citizens have a direct interaction with the state.

Finally, while this paper shows that systematic tax collection (by spurring citizens' involvement in the state's activities) can induce citizens to respond more positively to the requests of the state, the act of taxing citizens can also make them hostile to the state, and ultimately reduce compliance with the law (Besley et al., 2021). Future research on the conditions that make one of these two channels prevail will be important to inform policymakers on how to fine-tune tax enforcement policy.

References

- Abramitzky, Ran, Leah Boustan, and Katherine Eriksson. 2020. "Do Immigrants Assimilate More Slowly Today than in the Past?" American Economic Review: Insights, 2(1): 125–41.
- Acemoglu, Daron, Camilo García-Jimeno, and James A Robinson. 2015. "State Capacity and Economic Development: a Network Approach." *American Economic Review*, 105(8): 2364– 2409.
- Acemoglu, Daron, Giuseppe De Feo, Giacomo De Luca, and Gianluca Russo. 2022. "War, Socialism, and the Rise of Fascism: an Empirical Exploration." The Quarterly Journal of Economics, 137(2): 1233–1296.
- Alesina, Alberto, Bryony Reich, and Alessandro Riboni. 2020. "Nation-Building, Nationalism, and Wars." Journal of Economic Growth, 25(4): 381–430.
- Alesina, Alberto, Paola Giuliano, and Bryony Reich. 2021. "Nation-Building and Education." *The Economic Journal*, 131(638): 2273–2303.
- Bazzi, Samuel, Martin Fiszbein, and Mesay Gebresilasse. 2020. "Frontier Culture: the Roots and Persistence of "Rugged Individualism" in the United States." *Econometrica*, 88(6): 2329–2368.
- Becker, Sascha O., Katrin Boeckh, Christa Hainz, and Ludger Woessmann. 2016. "The Empire Is Dead, Long Live the Empire! Long-Run Persistence of Trust and Corruption in the Bureaucracy." *The Economic Journal*, 126(590): 40–74.
- **Beck Knudsen, Anne Sofie.** 2019. "Those Who Stayed: Selection and Cultural Change during the Age of Mass Migration." University of Copenhagen. Department of Economics. Discussion Paper 19-01.
- Berg, Heléne, Matz Dahlberg, and Kåre Vernby. 2019. "Post-WWI military disarmament and interwar fascism in Sweden." *Historical Methods: A Journal of Quantitative and Interdisciplinary History*, 52(1): 37–56.
- Besley, Timothy, Anders Jensen, and Torsten Persson. 2021. "Norms, Enforcement, and Tax Evasion." The Review of Economics and Statistics, 1–28.
- Besley, Timothy, and Torsten Persson. 2009. "The Origins of State Capacity: Property Rights, Taxation, and Politics." *American Economic Review*, 99(4): 1218–44.
- Besley, Timothy, and Torsten Persson. 2010. "State Capacity, Conflict, and Development." Econometrica, 78(1): 1–34.
- Blanc, Guillaume, and Masahiro Kubo. 2021. "Schools, Language, and Nations: Evidence from a Natural Experiment in France."
- Boehnke, Jörn, and Victor Gay. 2022. "The Missing Men: World War I and Female Labor Force Participation." *Journal of Human Resources*, 57(4): 1209–1241.

- Cameron, A. Colin, Jonah B. Gelbach, and Douglas L. Miller. 2008. "Bootstrap-Based Improvements for Inference with Clustered Errors." *The Review of Economics and Statistics*, 90(3): 414–427.
- Campante, Filipe, and David Yanagizawa-Drott. 2015. "The Intergenerational Transmission of War." National Bureau of Economic Research Working Paper 21371.
- Cantoni, Davide, Cathrin Mohr, and Matthias Weigand. 2019. "The Rise of Fiscal Capacity." CRC TRR 190 Rationality and Competition Rationality and Competition Discussion Paper Series 172.
- Cantoni, Enrico. 2020. "A Precinct Too Far: Turnout and Voting Costs." American Economic Journal: Applied Economics, 12(1): 61–85.
- Caprettini, Bruno, and Hans-Joachim Voth. 2023. "New Deal, New Patriots: How 1930s Government Spending Boosted Patriotism During World War II." The Quarterly Journal of Economics, 138(1): 465–513.
- Chen, Daniel L. 2016. "The Deterrent Effect of the Death Penalty? Evidence from British Commutations During World War I." Toulouse School of Economics (TSE) TSE Working Papers 16-706.
- Chilosi, David. 2007. "Old Wine in New Bottles: Civic Nation-Building and Ethnic Nationalism in Schooling in Piedmont, ca. 1700–1861." *Nations and Nationalism*, 13(3): 417–436.
- Colella, Fabrizio, Rafael Lalive, Seyhun Orcan Sakalli, and Mathias Thoenig. 2019. "Inference with Arbitrary Clustering." IZA Discussion Paper 12584.
- Commissariato Generale dell'Emigrazione. 1924. Il contributo Dato alla Vittoria dal Commissariato Generale dell'Emigrazione. Mobilitazione e Smobilitazione degli Emigrati Italiani in Occasione della Guerra, 1915-1922. Cartiere Centrali Roma.
- Corbetta, Piergiorgio, and Maria Serena Piretti. 2009. Atlante storico-elettorale d'Italia 1861-2008. Bologna, Zanichelli.
- Costa, Dora L., and Matthew E. Kahn. 2003. "Cowards and Heroes: Group Loyalty in the American Civil War*." The Quarterly Journal of Economics, 118(2): 519–548.
- Daly, Selena. 2021. "Emigrant Draft Evasion in the First World War: Decision-Making and Emotional Consequences in the Transatlantic Italian Family." *European History Quarterly*, 51(2): 170– 188.
- Dell, Melissa, Nathan Lane, and Pablo Querubin. 2018. "The Historical State, Local Collective Action, and Economic Development in Vietnam." *Econometrica*, 86(6): 2083–2121.
- **De Michelis, Giuseppe.** 1926. L'Emigrazione Italiana dal 1910 al 1923: Relazione Presentata a S.E. il Ministro degli Affari Esteri dal Commissario Generale dell'Emigrazione. Vol. 1, Edizioni del Commissariato Generale dell'Emigrazione.

- Dincecco, Mark, and Gabriel Katz. 2014. "State Capacity and Long-Run Economic Performance." The Economic Journal, 126(590): 189–218.
- **Dincecco, Mark, Giovanni Federico, and Andrea Vindigni.** 2011. "Warfare, Taxation, and Political Change: Evidence from the Italian Risorgimento." *The Journal of Economic History*, 71(4): 887–914.
- Dincecco, Mark, James Fenske, Anil Menon, and Shivaji Mukherjee. 2021. "Pre-Colonial Warfare and Long-Run Development in India." *The Economic Journal*, 132(643): 981–1010.
- Direzione Generale della Statistica. 1886. Risultati dell'Inchiesta sulle Condizioni Igieniche e Sanitarie nei Comuni del Regno - Parte II. Roma, Tipografia in San Michele di Reggiani e Soci.
- **Direzione Generale della Statistica.** 1887. *Bilanci Comunali per l'Anno 1884.* Roma, Stabilimento Tipografico dell'Opinione.
- **Direzione Generale della Statistica.** 1896. Bilanci dei Singoli Comuni Riassunti Secondo i Titoli Principali di Entrata e Spesa per l'Anno 1895. Roma, Tipografia Elzeviriana.
- **Direzione Generale della Statistica.** 1913. Annuario Statistico Italiano. Volume II 1912. Roma, Tipografia Bertero.
- Dube, Arindrajit, T. William Lester, and Michael Reich. 2010. "Minimum Wage Effects Across State Borders: Estimates Using Contiguous Counties." The Review of Economics and Statistics, 92(4): 945–964.
- Durante, Ruben, Nicola Mastrorocco, Luigi Minale, and Jr. Snyder, James M. 2023. "Unpacking Social Capital." National Bureau of Economic Research Working Paper 31083.
- Eandi, Giovanni. 1835. Statistica della Provincia di Saluzzo Volume II. Saluzzo, Domenico Lobetti-Bodoni.
- Elias, Norbert. 1994. The Civilizing Process. Oxford, Blackwell.
- Esposito, Elena, Tiziano Rotesi, Alessandro Saia, and Mathias Thoenig. 2023. "Reconciliation Narratives: The Birth of a Nation after the US Civil War." *American Economic Review*, 113(6): 1461–1504.
- FAO. 2015. "Global Agro-Ecological Zones."
- Fossati, Antonio. 1930. "Bilanci, Tributi Redditi e Valori negli Stati di Terraferma dalla Restaurazione all'Avvento di Carlo Alberto." *Rivista Internazionale di Scienze Sociali e Discipline Ausiliarie*, 1(5): 439–461.
- Fouka, Vasiliki. 2019. "Backlash: The Unintended Effects of Language Prohibition in U.S. Schools after World War I." *The Review of Economic Studies*, 87(1): 204–239.
- Fryer Jr, Roland G, and Steven D Levitt. 2004. "The Causes and Consequences of Distinctively Black Names." The Quarterly Journal of Economics, 119(3): 767–805.
- Gennaioli, Nicola, and Hans-Joachim Voth. 2015. "State Capacity and Military Conflict." The Review of Economic Studies, 82(4): 1409–1448.
- Giuliano, Paola, and Marco Tabellini. 2023. "The Seeds of Ideology: Historical Immigration and Political Preferences in the United States." Harvard Business School Working Paper 20-118.
- Heldring, Leander. 2021. "The Origins of Violence in Rwanda." *The Review of Economic Studies*, 88(2): 730–763.
- Heldring, Leander. 2023. "Bureaucracy as a Tool for Politicians: Evidence from Germany."
- Henn, Soeren J. 2022. "Complements or Substitutes? How Institutional Arrangements Bind Traditional Authorities and the State in Africa." *American Political Science Review*, 1–20.
- Ilari, Virgilio. 1990. Storia del Servizio Militare in Italia dal 1871 al 1918. CeMiSS.
- Johnson, Noel D. 2015. "Taxes, National Identity, and Nation Building: Evidence from France." University Library of Munich MPRA Paper 63598.
- Kaldor, Nicholas. 1963. "Taxation for Economic Development." The Journal of Modern African Studies, 1(1): 7–23.
- Keizer, Kees, Siegwart Lindenberg, and Linda Steg. 2008. "The Spreading of Disorder." Science, 322(5908): 1681–1685.
- Kleykamp, Meredith A. 2006. "College, Jobs, or the Military? Enlistment during a Time of War." Social Science Quarterly, 87(2): 272–290.
- Koenig, Christoph. 2023. "Loose Cannons: War Veterans and the Erosion of Democracy in Weimar Germany." *The Journal of Economic History*, 83(1): 167–202.
- Lax-Martinez, Gema, Dominic Rohner, and Alessandro Saia. 2022. "Threat of Taxation, Stagnation and Social Unrest: Evidence from 19th Century Sicily." Journal of Economic Behavior Organization, 202: 361–371.
- Levi, Margaret. 1996. "A State of Trust." EUI Working Paper 96/23, European University Institute.
- Lichter, Andreas, Max Löffler, and Sebastian Siegloch. 2020. "The Long-Term Costs of Government Surveillance: Insights from Stasi Spying in East Germany." Journal of the European Economic Association, 19(2): 741–789.
- Lowes, Sara, Nathan Nunn, James A Robinson, and Jonathan L. Weigel. 2017. "The Evolution of Culture and Institutions: Evidence from the Kuba Kingdom." *Econometrica*, 85(4): 1065–1091.
- Marciante, Gianni. 2023. "When Nation Building Goes South: Draft Evasion, Government Repression, and the Origins of the Sicilian Mafia." Working Paper.

Melograni, Piero. 1969. Storia Politica della Grande Guerra 1915-1918. Editori Laterza.

- Michalopoulos, Stelios, and Elias Papaioannou. 2014. "National Institutions and Subnational Development in Africa." The Quarterly Journal of Economics, 129(1): 151–213.
- Ministero di Agricoltura, Industria e Commercio. 1862. Statistica del Regno d'Italia. Popolazione. Censimento degli Antichi Stati Sardi (1° Gennaio 1858) e Censimenti di Lombardia, di Parma e di Modena (1857-1858). Torino, Stamperia Reale.
- Ministero di Agricoltura, Industria e Commercio. 1874. Popolazione Presente ed Assente per Comuni, Centri e Frazioni di Comune. Censimento 31 Dicembre 1871. Vol. 1, Roma, Stamperia Reale.
- Norsa, Paolo, and Mario Da Pozzo. 1961. Imposte e Tasse in Piemonte Durante il Periodo Cavouriano. Torino, Museo Nazionale del Risorgimento.
- Nunn, Nathan, and Leonard Wantchekon. 2011. "The Slave Trade and the Origins of Mistrust in Africa." *American Economic Review*, 101(7): 3221–52.
- **Oster, Emily.** 2019. "Unobservable Selection and Coefficient Stability: Theory and Evidence." Journal of Business & Economic Statistics, 37(2): 187–204.
- Özak, Ömer. 2010. "The Voyage of Homo-Economicus: Some Economic Measures of Distance." Department of Economics, Brown University.
- Özak, Ömer. 2018. "Distance to the Pre-Industrial Technological Frontier and Economic Development." Journal of Economic Growth, 23(2): 175–221.
- **Pinker, Steven.** 2011. The Better Angels of Our Nature: The Decline of Violence in History and its Causes. Penguin UK.
- **Plebano, Terenzio.** 1832. Statistica del Mandamento di Baldichieri, Provincia d'Asti. Tipografia Pomba.
- Plebano, Terenzio. 1836. Statistica del Mandamento di Riva presso Chieri. Torino, Tipografia Cassone, Marzorati, Vercellotti.
- **Pomeranz, Dina, and José Vila-Belda.** 2019. "Taking State-Capacity Research to the Field: Insights from Collaborations with Tax Authorities." *Annual Review of Economics*, 11: 755–781.
- Qian, Nancy, and Marco Tabellini. 2021. "Discrimination and State Capacity: Evidence from WWII U.S. Army Enlistment." National Bureau of Economic Research Working Paper 29482.
- **Regia Segreteria degli Affari Interni.** 1841. Relazione a S.M. sulla Situazione degli Istituti di Carità e Beneficenza dopo l'Editto 24 Dicembre 1836. Torino, Stamperia Reale.
- **Regno di Sardegna.** 1824. *Calendario Generale pe' Regii Stati Primo Anno.* Torino, Stamperia Vedova Pompa e Figli.

- Rei, Claudia. 2020. "Priests and Postmen: Historical Origins of National Identity." CAGE Working Paper 496.
- **Restrepo, Pascual.** 2015. "The Mounties and the Origins of Peace in the Canadian Prairies." MIT Department of Economics Working Paper 15-01.
- Ronconi, Juan Pedro, and Diego Ramos-Toro. 2023. "Nation-Building through Military Service." Working Paper.
- Rouanet, Louis, and Ennio E. Piano. 2023. "Drafting the Great Army: The Political Economy of Conscription in Napoleonic France." *The Journal of Economic History*, 83(4): 1057–1100.
- **Russo, Gianluca.** 2019. "Mass Media and Cultural Homogenization: Evidence from the Golden Age of Radio in the United States, 1920-1940."
- Santi, B. 1902. Dizionario dei Comuni del Regno d'Italia Secondo il Censimento del 10 Febbraio 1901. Milano, Ulrico Hoepli.
- Scott, James C. 2017. Against the Grain: A Deep History of the Earliest States. Yale University Press.
- Statistica del Regno d'Italia. 1865. Istruzione Primaria: Istruzione Elementare Pubblica per Comuni: Anno Scolastico 1862-1863. Modena, Tipografia di Antonio ed Angelo Cappelli.
- Statistica del Regno d'Italia. 1867. Elezioni Politiche e Amministrative. Anni 1865-66. Firenze, Tipografia Tofani.
- Statistica del Regno d'Italia. 1870. Gli Asili Infantili nel 1869. Firenze, Tipografia Tofani.
- **Stefani, Guglielmo.** 1855. *Dizionario Generale Geografico-Statistico degli Stati Sardi.* Cugini Pomba.
- Varnum, Michael E.W., and Shinobu Kitayama. 2011. "What's in a Name? Popular Names are Less Common on Frontiers." *Psychological science*, 22(2): 176–183.
- Weber, Eugen. 1976. Peasants into Frenchmen: the Modernization of Rural France, 1870-1914. Stanford University Press.
- Weigel, Jonathan L. 2020. "The Participation Dividend of Taxation: How Citizens in Congo Engage More with the State When it Tries to Tax Them*." The Quarterly Journal of Economics, 135(4): 1849–1903.
- Weigel, Jonathan L., and Elie Kabue Ngindu. 2023. "The Taxman Cometh: Pathways Out of a Low-Capacity Trap in the Democratic Republic of the Congo." *Economica*, 90(360): 1362–1396.

Figures



Figure 1: Italy, Piedmont and Tax Districts

(b)

Notes: Panel (a) shows a map presenting in darker blue the territories of the Sardinia Kingdom (formal name of the area mostly corresponding to Piedmont and the Sardinian island) as a part of Italy; panel (b) zooms into Piedmont and presents its towns (the blue polygons), its tax districts (delimited by red borders), tax districts' capitals (green dots) and in darker blue the towns of the province of Turin in 1917, where the citizens' of the dataset lived at the time of the war. The shapefile of towns is constructed as explained in B.1, using data from ISTAT; the assignment of towns to tax districts follows Stefani (1855).

Figure 2: Town-pairs fixed-effects and walking distances from tax collectors' residences: example



Notes: This map illustrates the identification strategy, showing two tax districts and highlighting one of the town-pairs used for identification. Blue circles represent tax district capitals (Corio and Rivara); red dots are the centroids of the other towns of the two districts; the red borders highlight two towns (Levone and Rocca Canavese) composing a matched pair of contiguous towns on the border; the two green lines represent the walking path between the town and its tax district capital (residence of a tax collector), computed as explained in 3.1. The shapefile of towns is constructed as explained in B.1, using data from ISTAT; the assignment of towns to tax districts follows Stefani (1855).



Figure 3: Balance of observable town-level characteristics

Notes: This figure shows β coefficients and 95% confidence intervals of separate regressions based on Equation 4.1, which controls for town-pairs and province fixed effects. Units of observation are towns × town-pair. The sample includes all the towns in Piedmont selected as explained in sections 4.2 and 4.3. The explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period. Dependent variables of the first set of regressions (in blue) are geographical or predetermined characteristics, namely: population of the town in 1824; logarithm of elevation of the town; distance from province capital, distance from division capital (*Circondario*), distance from Turin; distances from the Italian borders with Austria, France and Switzerland. Dependent variables of the second set of regressions (in red) are other town-level characteristics, namely: population in 1901; share of internal immigrants living in the town; share of citizens born in the town living elsewhere; indicator variables for the presence in the town of a post office, a train station, or a police station.



Figure 4: Distance from tax collector and municipal tax revenues, Ivrea and Saluzzo

Notes: This figure presents graphical evidence on the relationship between the (log) walking distance of a town from a former tax collector's residence in the 1814-1870 period (computed as explained in 3.1) and the amount of taxes raised on behalf of the town government before 1870, transformed using the inverse hyperbolic sine function. Tax data are from 1846 and 1831-1834, and pertain to towns of the divisions of Ivrea (1846) and Saluzzo (1831-34), selected as explained in sections 4.2 and 4.3. The binscatter shows the association between the two variables after controlling for town-pair fixed effects, province fixed effects, and geographical and pre-determined characteristics (see B.3 for their descriptions).



Figure 5: Distance from tax collector and share of draft evaders

Notes: This figure presents graphical evidence on the relationship between the (log) walking distance of a town from a former tax collector's residence in the 1814-1870 period (computed as explained in 3.1) and the share of draft evaders of the 1899 cohort born in the town. Data are from towns of the Turin sample, selected as explained in sections 4.2 and 4.3. The binscatter shows the association between the two variables after controlling for town-pair fixed effects, province fixed effects and geographical and predetermined characteristics (see Table 1 for their description).



Figure 6: Placebo: Effect of distance from judicial district capital

Notes: This figure presents graphical evidence on the relationship between the (log) walking distance of a town from a judicial district capital (computed as explained in 3.1) and the share of draft evaders of the 1899 cohort born in the town. Data are from towns of the Vicenza sample, selected as explained in sections 4.2, 4.3 and 6.3. The binscatter shows the association between the two variables after controlling for town-pair fixed effects, province fixed effects and geographical and pre-determined characteristics (see Table 3 for their description).





Notes: The figure presents graphical evidence on the relationship between (log) walking distance of a town from a former tax collector's residence in 1814-1870 and the amount of taxes raised by town governments both before and after the removal of tax districts in 1870. The outcome variable amount of taxes raised by town governments before 1870, transformed using the inverse hyperbolic sine function. Plotted coefficients and 95% confidence intervals are estimated using equation 7.1, which controls for year fixed effects, geographical and pre-determined characteristics (see Table 3 for their description), town-pairs and province fixed effects, and their interaction with a post-reform indicator variable (for the two years after the removal of tax districts). The coefficients plotted in blue are estimated using data from towns of the divisions of Ivrea and Saluzzo for which tax data before 1870 are available, while coefficients plotted in sections 4.2 and 4.3.

Tables

	ç	Share Draft Evader	s
	(1)	(2)	(3)
(log) Walking Dist. from Tax Collector	0.015^{**} (0.006)	0.026^{**} (0.011)	0.036^{***} (0.011)
DV Mean	0.0472	0.0472	0.0472
Town-pair FE	yes	yes	yes
Province FE	yes	yes	yes
No district capitals	yes	yes	yes
Geographical controls	no	yes	yes
Mean individual controls	no	no	yes
Town-level controls	no	no	yes
Towns in sample	135	135	135
Observations	232	232	232
R-squared	0.587	0.678	0.776

Table 1: Distance from former tax district capitals and share of draft evaders

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. In every column the dependent variable is the share of draft evaders born in a town in 1899 out of all the men born in 1899 in the town. The sample includes all the towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distance from Turin, population in 1824. Mean of individual controls are town-level means of individual characteristics as measured by military data, including height, literacy, being resident abroad, and having had an exemption for medical reasons. Town-level controls include population in 1901, distance from the closest police station, and indicator variables for the presence in the town of a post office, a police station, or a train station. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

			Draft Evade	r	
	(1)	(2)	(3)	(4)	(5)
Walking Dist. from Tax Collector, i.h.s. trans.	0.013*		0.014*	0.049***	0.054***
	(0.008)		(0.008)	(0.018)	(0.016)
Walking Dist. from Police Station, i.h.s. trans.	()	0.004**	0.004**	0.004*	0.007**
		(0.002)	(0.002)	(0.002)	(0.003)
DV Mean	0.0658	0.0658	0.0658	0.0658	0.0658
Town-pair FE	yes	yes	yes	yes	yes
Province FE	yes	yes	yes	yes	yes
No district capitals	yes	yes	yes	yes	yes
Geographical controls	no	no	no	yes	yes
Mean individual controls	no	no	no	no	yes
Town-level controls	no	no	no	no	yes
Individuals in sample	2112	2112	2112	2112	2112
Towns in sample	135	135	135	135	135
Observations	3,892	3,892	3,892	3,892	3,892
R-squared	0.051	0.053	0.053	0.067	0.076

Table 2: Distance from former tax district capitals, distance from police and likelihood toevade the draft

Notes: The table reports estimates from Equation 4.2 (which controls for town-pair and province fixed effects), showing estimated coefficient β and the estimated effect of distance from the closest police station of the town of residence of the individual. Units of observation are individuals \times town-pair. In every column the dependent variable is a binary variable indicating whether the individual evaded the draft, or not. The sample includes all the individuals born in towns in Piedmont selected as explained in section 4.3. The first explanatory variable is the inverse hyperbolic sine transformation of the walking distance between the town of birth of the citizen and the residence of the tax collector of the town in the 1814-1870 period (or district capital), computed as explained in section 3.1. The second explanatory variable is the inverse hyperbolic sine transformation of the walking distance of the town of residence of the drafted citizen from the closest police station, computed in the same way as the first explanatory variable. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (Circondario), distance from Turin, population in 1824. Mean of individual controls are town-level means of individual characteristics as measured by military data, including height, literacy, being resident abroad, and having had an exemption for medical reasons. Town-level controls include population in 1901 and indicator variables for the presence in the town of a post office, a police station, or a train station. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

		Share Draft Evader	s
	(1)	(2)	(3)
(log) Walking Dist_from District Capital	-0.007	-0.013	0 024
(log) Walking Dist. Hold District Capital	(0.025)	(0.034)	(0.025)
p-value	[0.770]	[0.703]	[0.352]
Wild Cluster Bootstrap-t p-value	[0.771]	[0.760]	[0.475]
DV Mean	0.0405	0.0405	0.0405
Town-pair FE	yes	yes	yes
Province FE	yes	yes	yes
No district capitals	yes	yes	yes
Geographical controls	no	yes	yes
Mean individual controls	no	no	yes
Town-level controls	no	no	yes
Towns in sample	55	55	55
Observations	100	100	100
R-squared	0.634	0.808	0.947

Table 3: Placebo test: Distance from district capitals, and share of draft evaders in
Vicenza

Notes: The table reports β coefficients from Equation 4.1 estimated on the Vicenza sample (men drafted in the province of Vicenza), which controls for town-pair and province fixed effects. Units of observation are towns \times town-pair. In every column the dependent variable is the share of draft evaders born in a town in 1899 out of all the men born in 1899 in the town. The sample includes all the towns out of Piedmont selected as explained in sections 4.3 and 6.3. The main explanatory variable is the logarithm of the walking distance between a town and the location of a minor court that had jurisdiction over the town (or district capital), computed as explained in section 3.1. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (Distretto), distance from Turin, imputed population in 1843-44. Mean of individual controls are town-level means of individual characteristics as measured by military data, namely being resident abroad and having had an exemption for medical reasons. Town-level controls include population in 1901, distance from a police station, and indicator variables for the presence in the town of a post office, a police station, or a train station. Standard errors in parentheses are clustered at the tax district and district-border level. Square brackets indicate p-values from baseline specification and from wild cluster bootstrap (clustering in both cases at the tax district and district-border level). *** p<0.01, ** p<0.05, * p<0.1.

		Draft Evader	
	(1)	(2)	(3)
(log) Walking Dist. from Tax Collector	0.069	0.251**	0.425***
	(0.045)	(0.117)	(0.104)
p-value	[0.132]	[0.039]	[0.000]
Wild Cluster Bootstrap-t p-value	[0.084]	[0.098]	[0.052]
DV Mean	0.0103	0.0103	0.0103
Residence FE	yes	yes	yes
Town-pair FE	yes	yes	yes
Province FE	yes	yes	yes
No district capitals	yes	yes	yes
Geographical controls	no	yes	yes
Mean individual controls	no	no	yes
Town-level controls	no	no	yes
Individuals in sample	180	180	180
Towns in sample	67	67	67
Observations	291	291	291
R-squared	0.430	0.585	0.649

Table 4: Distance from former tax district capitals and likelihood to evade the draft:sample of migrants

Notes: The table reports β coefficients from Equation 4.2 (which controls for town-pair and province fixed effects) estimated on the sample of internal migrants, and including town-of-residence fixed effects in each specification. Units of observation are individuals \times town-pair. In every column the dependent variable is a binary variable indicating whether the individual evaded the draft, or not. The sample includes all the individuals born in towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between the town of birth of the citizen and the residence of the tax collector of the town in the 1814-1870 period (or district capital), computed as explained in section 3.1. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (Circondario), distance from Turin, population in 1824. Mean of individual controls are town-level means of individual characteristics as measured by military data, including height, literacy, being resident abroad, and having had an exemption for medical reasons. Town-level controls include population in 1901, distance from the closest police station, and indicator variables for the presence in the town of a post office, a police station, or a train station. Standard errors in parentheses are clustered at the tax district and district-border level. Square brackets indicate p-values from baseline specification and from wild cluster bootstrap (clustering in both cases at the tax district and district-border level). *** p<0.01, ** p<0.05, * p<0.1.

	Turnout Loo	cal Elections	Turnout National Elections 1913		
	18	65			
	(1)	(2)	(3)	(4)	
(log) Walking Dist. from Tax Collector	-0.068**	-0.091**	-0.043*	-0.037*	
	(0.031)	(0.034)	(0.023)	(0.021)	
DV Mean	0.396	0.396	0.683	0.683	
Town-pair FE	yes	yes	yes	yes	
Province FE	yes	yes	yes	yes	
No district capitals	yes	yes	yes	yes	
Geographical controls	no	yes	no	yes	
Towns in sample	135	135	135	135	
Observations	232	232	232	232	
R-squared	0.612	0.717	0.664	0.783	

Table 5: Distance from former tax district capitals and voter turnout, 1865 and 1913

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. The sample includes all the towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. In columns (1) and (2) the dependent variable is voter turnout in local elections, computed using data from Statistica del Regno d'Italia (1867) as the ratio between voters and eligible voters in 1865 local elections. In columns (3) and (4) the dependent variable is voter turnout in national elections, using data from the historical archive of the Italian Chamber of Deputies, Corbetta and Piretti (2009) and Direzione Generale della Statistica (1913) as the ratio between voters in 1913 national elections. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (*Circondario*), distance from Turin. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

	Expenditure on Charity and Hospitals		Expend Primary	Expenditure on Exp Primary Schools Ki		Expenditure on Kindergartens		Presence of Sewage Systems		Number Public-Paid Physicians	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	
(log) Walking Dist. from Tax Collector	0.036 (0.488)	-0.374 (0.640)	-0.440 (0.279)	-0.136 (0.211)	-0.059 (0.103)	-0.066 (0.303)	-0.005 (0.005)	0.006 (0.014)	0.034 (0.082)	0.039 (0.089)	
DV Mean	3.893	3.893	7.475	7.475	0.571	0.571	0.0129	0.0129	0.388	0.388	
Town-pair FE	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	
Province FE	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	
No district capitals	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	
Geographical controls	no	yes	no	yes	no	yes	no	yes	no	yes	
Towns in Sample	135	135	135	135	135	135	135	135	135	135	
Observations	232	232	232	232	232	232	232	232	232	232	
R-squared	0.609	0.705	0.661	0.842	0.594	0.692	0.493	0.578	0.678	0.753	

Table 6: Distance from former tax district capitals and public good provision

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. The sample includes all the towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. In columns (1) and (2) the dependent variable is total revenues (per town) of charity institutes funded by the Piedmontese government in 1840. In columns (3) and (4) the dependent variable is the amount of public expenditure on town's primary schools in 1862-1863. In columns (5) and (6) the dependent variable is the the amount of public expenditure on town's kindergartens in 1862-1863. In columns (7) and (8) the dependent variable is the number of physicians paid by towns' governments. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distance from Turin. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

	Share Medals of Honor		Share Ki	ngs' Names
	(1)	(2)	(3)	(4)
log) Walking Dist. from Tax Collector	-0.011*	0.007	0.006	0.005
	(0.006)	(0.008)	(0.005)	(0.006)
DV Mean	0.0425	0.0425	0.0185	0.0185
Town-pair FE	yes	yes	yes	yes
Province FE	yes	yes	yes	yes
No district capitals	yes	yes	yes	yes
Geographical controls	no	yes	no	yes
Towns in Sample	134	134	135	135
Observations	224	224	232	232
R-squared	0.585	0.732	0.538	0.635

Table 7: Distance from former tax district capitals, patriotism and trust

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. The sample includes all the towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. In columns (1) and (2) the dependent variable is the share of soldiers born in the town who received a gold, silver or bronze Medal of Honor, computed on the sample of soldiers who died during World War I. In columns (3) and (4) the dependent variable is the share of male citizens born in the town named after a Savoy King. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (*Circondario*), distance from Turin. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

Appendix

A Additional figures

Figure A.1: Towns of birth of soldiers in Turin full sample and Turin regression sample



Notes: Panel (a) shows a map presenting in (in brown) all the towns in which at least one drafted citizen of the Turin sample was born. Panel (b) shows the towns of birth of citizens in the regression sample, selected as explained in sections 4.2 and 4.3. The shapefile of towns is constructed as explained in B.1, using data from ISTAT.

Figure A.2: Towns of birth of soldiers in Vicenza full sample and Vicenza regression sample



Notes: Panel (a) shows a map presenting in (in brown) all the towns in which at least one drafted citizen of the Vicenza sample was born. Panel (b) shows the towns of birth of citizens in the regression sample, selected as explained in sections 4.2 and 4.3. The shapefile of towns is constructed as explained in B.1, using data from ISTAT.

B Data appendix

B.1 Map of towns in the Turin sample and in the Vicenza sample

I build the dataset for the main analysis starting from a map of Piedmont before the Italian unification in 1861. I create this map combining complete lists of towns⁶⁰ belonging to Piedmont⁶¹ with a shapefile of Italian towns in 2011. I use towns' and provinces' names to link towns in Piedmont to Italian towns in 1861, and then track changes in administrative boundaries of Italian towns to link them to the 2011 shapefile.⁶² I compute the distance from a tax collector's residence using centroids of Italian towns in 2011.

I follow the same procedure when building the database for the Vicenza sample, using all the Italian towns not belonging to Piedmont before $1861.^{63}$

B.2 Military data

I use data on drafted Italian male citizens that were living in the province of Turin in 1917 and were called to serve in the Army in that year. This sample includes the overwhelming majority of men born in 1899 living in the province of Turin at the time of the war and a limited number of older men who were exempted from the draft in previous years. I obtain my military data by collecting and digitising the lists of drafted men for all the towns belonging to the province of Turin in 1917 (around 5250 documents), publicly accessible at the Turin State Archive. For the city of Turin I collected information on all drafted citizens that were born in another town (i.e. internal migrants) and for a subsample of men born in Turin;⁶⁴ for all the other towns of the province (442 in 1917, 388 in 2011) I collected complete draft lists.

For the Vicenza sample, I collected data on all the men of the 1899 cohort living in the province of Vicenza by the time of the war; I obtain these data from ARSAS.⁶⁵

B.3 Variables' construction

Measures of historical fiscal capacity: distance from tax collectors' residence. I compute the distance of a town from the residence of its tax collector (using ArcGIS) as the natural logarithm of the optimal walking route in meters, computed using as impediment both elevation, weather patterns and soil conditions (using data from Özak, 2010 and Özak, 2018). Distances are computed between centroids of towns using ArcGIS. For the Vicenza sample, I use the same measure computed between a town and a (judicial) district capital.

⁶⁰Lists of towns in Piedmont are from 1824, 1848, 1901 and 2011.

⁶¹ formally, the mainland of the Sardinia Kingdom

 $^{^{62}\}mathrm{I}$ track changes from 1861 to 2011 using data from Elesh (www.elesh.it).

⁶³In this case, I build the dataset using lists of Italian towns from 1871, 1901 and 2011.

 $^{^{64}}$ A limited number of drafted men were not included in the sample because of illegibility of the documents. Note that men born in Turin, a city that was hosting a tax collector's residence, would never enter the analysis sample as described in section 4.3.

⁶⁵Associazione Recupero Salvaguardia Archivi Storici, (accessible at www.arsas.org).

As an alternative measure, I use the natural logarithm of the Euclidean distance between a town and its tax collector's residence, and the (natural logarithm of) the cost distance in days using again data from the Human Mobility Index by Özak (2010).

Draft evasion. I build my measures of draft evasion in the following way: first, I detect whether the individual was ever classified as a draft evader; second, I inspect whether the sanction of draft evasion was revoked later,⁶⁶ and classify as draft evaders only individuals for which the sanction was never revoked.

When conducting the analysis at the town \times town-pair, I build my outcome variable as the share of draft evaders (classified as described above) in my lists born in the town over the total number of individuals in my lists born in the town. When conducting the analysis at the individual \times town-pair, my outcome variable is a binary indicator for whether the individual was classified as a draft evader or not. In the Vicenza sample, I use a binary indicator for whether a citizen is a draft evader according to the classification provided by ARSAS, for which being enlisted, exempted or evader are mutually exclusive categories.⁶⁷

Geographical variables

Elevation. I obtain elevation data of Italian towns in 2011 from ISTAT, and link them to the shapefile of Italian towns in 2011. I compute the natural logarithm of elevation and the natural logarithm of the standard deviation of elevation within the town.

Area. I obtain data on the area of towns in 2011 (surface in squared meters) from ISTAT, link them to the shapefile of Italian towns in 2011, and compute the natural logarithm of the town area.

Land suitability. I obtain data on land suitability from FAO (2015). This data is defined on a 9.25 x 9.25 km grid covering the entire planet. I join the rasters to the 2011 shapefile of Italian towns and assign to each town the potential yields with low-level inputs of the grid cells falling inside the town limits. I use land suitability for wheat, buckwheat, pasture grasses, barley, potato, oat, rice, and maize.

Coordinates. I compute the latitude and longitude of towns using the coordinates of their centroids, measured in degrees in the WGS84 UTM32N coordinate system.

Distances from country's borders. I compute the distances of a town's centroid from the Italian borders using ArcGIS.

Distance from Turin. I compute the distances of a town's centroid from the centroid of Turin using ArcGIS.

Distance from province capital. I compute the walking distance of a town's centroid from the centroid of its province capital (with elevation as an impediment) using ArcGIS.

Distance from division capital. I compute the walking distance of a town's centroid from the centroid of its *Circondario* capital (with elevation as an impediment) using ArcGIS.

⁶⁶For example, if the citizen showed up shortly later on, or was enlisted in another town, or joined the Navy.

⁶⁷In the classification by ARSAS, drafted citizens can only be "Enlisted", "Evaders" or "Exempt", without any motivation for the assignment to a category.

Provinces and divisions: 1917. I assign towns to their 1917 province.⁶⁸ For the Turin sample I assign towns to the division (*Circondario*) it belonged to before the Italian unification, which remained mostly unchanged until 1917. As the province of Vicenza did not have divisions like in the rest of Italy, I assign it to its *Distretto* (an administration similar to the division).

Tax and judicial districts. I assign towns in the Turin sample to the tax districts it belonged to in 1848 (according to Stefani, 1855). Towns in the Vicenza sample are assigned according to the districts they belonged to in 1871, shortly after they got annexed to the Italian Kingdom.

Individual characteristics

Illiteracy. I compute the share of illiterate male citizens in a town as the share of citizens in the 1917 military lists being unable or barely able to write (out of the total number of citizens in the 1917 lists born of the town).

Height. I compute the average height of male citizens as the average height for citizens present in the 1917 lists and born in the town who underwent the medical examination.

International migrant. I define the share of international migrants in a town as the share of citizens born in the town in the 1917 lists who were declared as being residents abroad in the military lists.

Exempted. I compute the share of citizens exempted for medical reasons in a town (out of the total number of citizens born in the town in the 1917 lists) by looking at the share of citizens who got any type of medical exemption (even temporary ones), and that in the Turin sample is not mutually exclusive with enlistment or draft evasion. For the Vicenza sample, I use the classification by ARSAS, for which being enlisted, exempted or evader are mutually exclusive categories.

$Town-level\ characteristics$

Local taxes in Piedmont. I obtain data on taxes raised by the tax collector on behalf of the town for the divisions (*circondari*) of Ivrea and Saluzzo. Data for the division of Ivrea are kept by the Turin State Archive and pertain to taxes collected in 1846; data from the division of Saluzzo come from Eandi (1835) and pertain average taxes collected between 1831 and 1834. The amount of taxes raised includes both direct and indirect taxes. For each town, I compute the inverse hyperbolic sine transformation of taxes raised, in absolute numbers and per capita (using data on population in 1847 for the Ivrea division, and in 1832 for the Saluzzo division).

Local tax revenues, 1884. From Direzione Generale della Statistica (1887) I collect data on the total amount of taxes collected by town governments in 1884.

Local tax revenues, 1895. From Direzione Generale della Statistica (1896) I collect data

⁶⁸Provinces of the towns in my sample did not change from the Italian unification until 1917.

on the total amount of taxes collected by town governments in 1895.

Population in 1824. I obtain the population for each town from the Turin sample in Piedmont in 1824 from Regno di Sardegna (1824), and link it to the 2011 shapefile as described in section B.1.

Population in 1843-1844. I compute the population for each town from the Vicenza sample in 1843-44 in the following way: first, I obtain population in 1871 (the first Italian census including the towns of the Veneto region) from Ministero di Agricoltura, Industria e Commercio (1874); second, I obtain *distretto*-level population in 1843-44 (the oldest census with such level of detail) from Ministero di Agricoltura, Industria e Commercio (1862) and I impute the population of a town in 1843-44 based on 1871 data and on the population growth of its *distretto* between 1843-44 and 1871. I then link the imputed population to the 2011 shapefile as described in section B.1.

Population in 1881. I obtain the population in 1881 for each town in the Turin, Vicenza, Ivrea and Saluzzo samples from Direzione Generale della Statistica (1886).

Population in 1901. I obtain the population in 1901 for each town in the Turin and the Vicenza samples from Santi (1902), and link it to the 2011 shapefile as described in section B.1.

In migration. I measure the share of internal migrants living in the town of birth of a drafted citizen as the share of residents in the town that I observe in the 1917 lists that were born in another town.

Out migration. I measure the share of internal migrants who migrated from the town of birth of a drafted citizen as the share of citizens I observe in the 1917 lists that were born in the town but had their residence in another town in 1917.

Presence of post office. I obtain information on the presence of a post office in a town from Santi (1902).

Presence of train station. I obtain information on the presence of a train station in a town from Santi (1902).

Presence of police station. I obtain information on the presence of a police station (more precisely, a *Carabinieri* station) in a town from Santi (1902).

Voter turnout in local elections. I compute voter turnout in local elections as the ratio between actual voters and eligible voters in 1865 elections for the city council, using town-level data from Statistica del Regno d'Italia (1867).

Voter turnout in national elections. I compute voter turnout in national elections in each town as the ratio between actual voters and *de facto* eligible voters in 1913 elections for the national Parliament, building the *de facto* franchise as explained in G.1 and using precinct-level data (aggregated at the town-level) from Corbetta and Piretti (2009), Direzione Generale della Statistica (1913) and newly hand-collected data on eligible voters from the historical archive of the Italian Chamber of Deputies.

Walking distance to the closest police station. I compute the walking distance to the closest police station as the inverse hyperbolic sine transformation of the optimal walking

route in meters, computed using as impediment both elevation, weather patterns and soil conditions (using data from Özak, 2010). Distances are computed using ArcGIS between the centroid of the town of residence of the drafted citizen and the centroid of the of town hosting the closest police station. I match towns to the closest police station using the centroids of all the police stations in Italy in 1901.

Public spending on primary schools. I obtain data on town-level spending in 1862-1863 from Statistica del Regno d'Italia (1865). I use the total amount of public funds received by primary schools from both central, provincial and town governments; in 1863, schools were almost exclusively funded by towns.

Public spending on kindergartens. I obtain data on town-level public spending on kindergartens in 1869 from Statistica del Regno d'Italia (1870). I use the total amount of public funds received by kindergartens from both central, provincial and town governments. **Presence of sewage systems in the town.** I use information on the presence of sewage systems in a town from Direzione Generale della Statistica (1886). I build an indicator variable equal to 1 when the presence of sewage systems is noted as "yes", "generally yes" or if a third or more of the streets are reported to have sewage systems.

Number of public-paid physicians in the town. From Direzione Generale della Statistica (1886) I obtain information on the number of doctors working in a town paid by the town's government to treat citizens free of charge.

Spending on charity institutions. I obtain data on spending on charity institutions from Regia Segreteria degli Affari Interni (1841). The variable concerns the total revenues of the charity institutions (including hospitals and orphanages) that were funded by the central government in 1840.

Medals of Honor. I use data from ISTORECO⁶⁹ on the casualties of Italian soldiers during WWI, and whether they obtained a Medal of Honor (gold, silver or bronze ones). For each town in my sample, I compute the share of dead soldiers born in that town who obtained a Medal of Honor from the Italian Army during World War I. Given the low number of medals, I build this variable using men from all the birth cohorts who died during WWI.

Share of citizens named after Kings. I use the first names of drafted citizens in the Turin sample to build the share of male citizens named after a king out of the total number of citizens drafted in the town. I classify a citizen as having a king's name if his first or middle names include "Vittorio", "Emanuele" or "Umberto", that is, the name of the king in 1899 or names of the previous king (Vittorio Emanuele II, first king of Italy).

 $^{^{69}\}mathrm{Data}$ are accessible at www.cadutigrandeguerra.it.

NASCITA E RESIDENZA Sebastia Deiro Me 216 1899 Luci Astonust Circondar names in Salto R.J. Classi di Carin. 1399 St Sallo 11 Crarotte não D Gircondari La DI 2 an 28 Tassi di 4

Figure B.1: Draft list of the Italian Army: Example

Notes: This figure presents an example of the original data collected in the Turin State Archive from the draft lists of the Italian Army. This example shows information for four drafted citizens born (and living in 1917) in the town of Salto. The first three columns contain: the name and surname of citizens; their town of birth, dates of birth, and names of their parents; personal information including height and literacy level. The box on the right contains accounts of the enlistment process of citizens.

	Regression sample					Full sample				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Obs.	Mean	S.D.	Min.	Max.	Obs.	Mean	S.D.	Min.	Max.
Panel A: town-level variables										
Share of draft evaders	135	0.05	0.08	0.00	0.50	660	0.04	0.08	0.00	1.00
Walking dist. from tax collector's residence (metres)	135	7669.02	4921.76	604.00	36006.00	663	4693.86	4411.95	0.00	36006.00
log walking dist. from tax collector's residence	135	8.79	0.56	6.40	10.49	663	8.62	0.50	6.40	10.49
log Elevation	135	6.28	0.89	4.75	7.93	663	6.09	0.87	4.21	7.93
log Area	135	2.71	0.96	0.71	5.35	663	2.86	0.90	0.45	5.36
Latitude	135	45.27	0.32	44.59	45.91	663	45.14	0.38	43.90	46.24
Longitude	135	7.73	0.32	6.95	8.54	663	7.80	0.45	6.73	9.91
Dist. from Division Capital (km)	135	21.11	13.20	2.78	62.64	663	18.51	11.39	0.00	62.64
Dist. from Province Capital (km)	135	49.72	28.75	8.68	151.87	663	46.13	28.14	0.00	151.87
Dist. from Turin (km)	135	51.41	29.40	8.68	151.87	663	59.30	36.96	0.00	262.20
Dist. from Austrian border (km)	135	227.59	23.72	183.13	281.96	663	228.61	32.67	132.33	310.03
Dist. from French border (km)	135	49.40	22.59	3.22	99.00	663	53.48	29.98	0.96	176.41
Dist. from Swiss border (km)	135	72.38	34.61	4.29	148.96	663	87.10	40.61	2.67	226.09
Population 1824	135	1406.34	1219.47	262	12172	663	2389.42	4821.35	45	101363
Population 1901	135	1774.22	1169.05	167	7116	663	3848.82	13984.88	62	329691
Presence of train station	135	0.20	0.40	0.00	1.00	663	0.30	0.46	0.00	1.00
Presence of post office	135	0.56	0.50	0.00	1.00	663	0.68	0.47	0.00	1.00
Presence of police station	135	0.14	0.35	0.00	1.00	663	0.31	0.46	0.00	1.00
Distance from police station (metres)	135	4923.42	3824.01	0.00	22838.00	663	3590.77	3261.70	0.00	22838.00
Out migration	135	1.73	2.29	0.00	12.00	663	2.77	6.82	0.00	147.00
In migration	135	1.29	2.50	0.00	19.00	663	1.30	3.88	0.00	64.00
Observations per town (town-pairs)	135	1.72	1.03	1	6	663	4.95	2.35	1.00	21.00
Average height	135	164.05	3.44	156.50	178.00	656	163.79	4.17	139.33	181.00
Share illiterate	135	0.02	0.06	0.00	0.40	661	0.04	0.11	0.00	1.00
Share exempted citizens	135	0.08	0.12	0.00	0.67	661	0.08	0.16	0.00	1.00
Average share of migrants abroad	135	0.01	0.03	0.00	0.14	661	0.01	0.03	0.00	0.50
Panel B: individual-level variables										
Height	1937	163.59	7.25	125	188	11539	163.71	7.40	105	195
Illiterate	1937	0.03	0.18	0.00	1.00	11539	0.04	0.20	0.00	1.00
Exempted	2112	0.09	0.28	0.00	1.00	13004	0.07	0.26	0.00	1.00
Migrant abroad	2112	0.01	0.10	0.00	1.00	13004	0.01	0.10	0.00	1.00
Panel C: tax district-level variables										
Number of towns	77	5.30	2.46	2	13	211	4.87	2.53	1.00	15.00
Total population (1824)	77	8691.90	4004.89	3639	32425	211	9552.51	7886.61	2793	101363

Table B.1: Summary statistics

Notes: The table presents summary statistics for variables used in the main analysis. Variables' construction is described in appendix B.3. Columns (1) to (5) present summary statistics from towns of the Turin regression sample, selected as explained in sections 4.3, and individuals living in these towns. Columns (6) to (610 present summary statistics from all the Piedmontese towns of birth of the drafted citizens in the Turin sample. Panel A presents statistics for town-level outcomes. Panel B presents statistics for individual-level outcomes: note that two variables (height and illiterate status) have missing observations, as they are missing for most draft evaders; for this reason, as explained in appendix B.3, in the analysis I use the town-level average of these variables based on non-missing observations. Panel C reports statistics on the average number of towns in a tax district and its total population.

\mathbf{C} Tax districts and tax collectors: anecdotal evidence

Figure C.1: 1801 Law: the creation of districts

13 brumajo (4.9.bre 1801) di prima instanza del circondario, che sarà tenuto di darla in mar-gine della spedizione, che le sarà presentata (art. 6 ibid.). Tit. III. De' tribunali di pace. §. 1. Del loro numero, e del loro circondario. 25. Vi sono per tutto il territorio subalpino, inaddietro piemon-tese, o 27 divisione militare, 197 tribunali di pace (estratto del decr. de' consoli 27 vendemm. ultimo). 26. I circondari de' tribunali di pace si regolano, purchè le lor-calità non vi arrecano ostacoli, sulle basi combinate della popo-lazione, e dell'estensione territoriale, nelle proporzioni seguenta (art. 21.8 piov an. 9, portante riduzione de' trib. di pace sin Francia). 27. La popolazione mediocre d'un circond. di trib. di pace sar. 27. La popolazione mediocre d'un circond, di trib. di pace sara 27. La populazione mediocre a un circona, di trib. di pace sara, di iomila abitanti. Il circond. non potrà abbracciarne più di quin-dici mila (art. 3 ibid.). 28. La mediocre estensione territoriale del circond. sarà di 250 bilomatri cunderti esten ione netrà comprenderne più di care est

28. La mediocre estensione territoriale dei circond. sarà di 25 kilometri quadrati; essa non potrà comprenderne più di 375, nè meno di 125 (ibid. art. 4). Nota. Il kilometro è una distanța, o misura itineraria di centa metri, ed il metro ha una lunghezza di tre piedi, undici llace, qua-rantaquattro centesmi, ciò che forma a un dipresso il pàsso. (29. Nulladimeno, ed allorché in una estensione territoriale mi-tora di int kilometri auderii esisterà una podelatione cumptione cumptione a

arts vila abitanti, la composizione de' circond. si farà in seguito alla sola base della popolazione (art. 5 ibid.). 30. Il territorio attuale delle piccole città, borghi e villaggi non potrà essere rotto, ne diviso, di maniera che parte ne sia data ad un circondario, se narte ad un altro. Diserco territorio està data ad

un circondario, e parte ad un abro. Questo territorio sarà conser-vato nella sua integrità, e situato in un solo e medesimo circondario di tribunale di pace (art. 6 ibid.).

31. La regola enunciara nel precedente articolo non è applica-bile alle comuni, che dalla loro propria popolazione avrebbero dritto allo stabilimento di più circondarj di tribunali di pace nel boro seno (art. 7 ibid.). 32. In ogni circond. di tribunale di pace, formato dalla riu-

nione di più comuni, il governo designerà quella, che, tanto pen ragione di sua centralità, che per rapporto alle sue relazioni cogli altri comuni del medesimo circond., ne sarà il capo-luogo (att. 8 ib.) 33. I circondari de' trib. di pace sono fissari conformemente alle stato inserto nel decr. de consoli 27 vendemmiaio,

Notes: Excerpt from the 1801's law "Règles générales sur l'administration de la Justice et sur l'organisation des Tribunaux dans le Piémont". Highlighted text: "The average population of a district will be of 10.000 citizens. The district could not have more than 15.000 citizens. [...] In the district of each JP, composed of a group of towns, the government will choose the town that, because of its centrality and of the relationships it has with the other towns of its district, will be the district capital."

Figure C.2: Debates from the Piedmontese Parliament: behaviour of tax collectors

BOTTA. Il discorso testà tenuto sul servizio degli esattori mi porge l'opportunità di volgere una preghiera al ministro delle finanze. Lo credo che non siasi derogato ad un provvedimento ministeriale, secondo il quale gli intendenti sono obbligati di fissare entro il mese, due o più giorni nei quali gli esattori, massime dei piccoli mandamenti, son tenuti di recarsi nei comuni onde far l'esazione. Questo servizio, o non si fa, o si fa irregolarmente. Io pertanto prego il ministro delle finanze acciò voglia vegliare onde gli intendenti facciano si che si dia esecuzione al provvedimento ministeriale di cui ho fatto cenno.

Notes: Excerpt from transcripts of the session of the Piedmontese House of Representatives (*Camera dei Deputati del Parlamento Subalpino*) held on June 1st, 1854. The excerpt is a short speech addressed by M.P. Luigi Botta to Prime Minister Cavour regarding tax collector's compliance with the guidelines they out to follow. Highlighted text: "If I understand correctly, it should still be in place a ministerial order for which the province governors [*Intendenti*] have to choose two or more days per month when the tax collectors, especially in small districts, are required to visit the towns of the district so to collect the taxes. This service either is not performed at all, or it is performed irregularly."

Figure C.3: Considerations on the allocation of towns to tax districts

Ha	Baldichieri	giornate	1,346	99
33	Cantarana	20	2,555	6
33	Castellero	33	1,209	8o
n	Monale	33	2,080	33
30	Settime	n	1,860	33
20	Tigliole	22	4,075	62
30	Villafranca	33	4,000	33
			17,127	47

8

L'estimo censuario si eseguì a Tigliole, e non altrove, nè qui però, nè a Villafranca evvi mappa, e regna in conseguenza grande confusione nelle proprietà fondiarie, sorgente di rustici alterchi, di litigi, d'usurpazioni.

Baldichieri meno elevato sul livello marittimo è il capoluogo della Regia Giudicatura. La forma di questa è alquanto bizzarra; Settime 5 miglia lontano da detto luogo, collocato tra li Mandamenti d'Asti, e di Montechiaro, presenta, dirò così, l'idea d'un'isola mandamentale, o se m'è lecito altro paragone, sembra un capo staccato dal corpo. La vicinanza a quest'ultimo paese, l'abitudine del popolo di settimanamente recarvisi al mercato stabilisce un tale rapporto con quel capo-luogo, che non esiste, nè può formarsi con Baldichieri. A questo proposito giova una riflessione. Lo scopo del Regio Governo nell'organizzazione de'Mandamenti fu certamente di rendere propta l'amministrazione della giustizia, facile l'accesso al tribunale nel minimo tempo, col minimo incomodo de'litiganti, colla minima spesa. Ora succede per appunto tutto l'opposto, laddove,

1.º Il paese assai dista dal capo-luogo.

2.º Non evvi con esso alcuna relazione commerciale.

3.º Manca una diritta strada, che vi conduca.

4.º Mancano opportuni ponti sulle acque intermedie.

Quindi osservo, che, se nella circoscrizione de' Mandamenti si vedono alcuni composti di 5, 10, ed anche 20, e più terre ; altri di 4, 3, ed anche di 2 soltanto, senza, che vi si scorga altronde alcuna proporzione di popolazione, od altra, deve almeno supporsi, che in ciò il Regio Governo una certa utilità pubblica ravvisò, dunque conchiuderò io, se questa stessa utilità richiede, che un paese sia piuttosto membro d'uno, che d'un altro Mandamento, frutterà incessante danno ogni ritardo a staccarlo dall'uno per incorporarlo coll'altro, consultandosi in questa organizzazione ben più la posizione rispettiva sotto li rapporti agrari, commerciali, e statistici d'ogni specie, che una mal intesa simetria, centralità, ammasso di popolazione, e che so io; il Mandamento di cui si tratta, non è il solo in questa situazione, ed è per ciò stesso, che ho fatto una breve digressione a tale riguardo.

Non parmi fuor di proposito il dare qui una succinta idea di ciascun paese su ciò, che non potrebbe in altro luogo di questo scritto essere trattato.

Baldichieri. Occupa il maggior abitato la falda della collina, ed il sottoposto piano; gli sovrasta l'avanzo d'antico diroccato castello; compreso egli tra le terre del capitanato Astese fu soggetto immediatamente alla Real Casa di Savoja, avente in sè la nomina del Pretore con prima, e seconda cognizione; il suo statuto scritto in latino a caratteri gotiei, di capi 195, porta la data del 1 aprile 1402. La Comunita sin dal 1534 aveva un pedaggio, la cui casa laterale allo stradone Reale ritiene ancora il nome; aveva pure sin dal 1648 il molino sulla bealera discorrente dal confine Monalese. La Regia Posta de 'cavalli vi csiste dal 1772; fu questo villaggio quattro volte saccheggiato circa il 1643; il suo catastro è del 1770, ed ha pur anco una mappa figurata; quasi della stessa data è il pubblico orologio; evvi in fine sonoro concerto di cinque campane di

Notes: Excerpt from the preface to Statistica del Mandamento di Baldichieri, by Plebano (1832), talking about the shape of the Baldichieri district, the location of the town of Settime (belonging to the district), and the inexplicable reason why Settime was assigned to the Baldichieri district. Highlighted text: "The shape of such a district [Baldichieri] is quite strange; the town of Settimo, located between the districts of Asti and Montechiaro, is 5 miles from such a place [Baldichieri], looking like an "island" of the district [of Baldichieri], or even a head detached from its body. The proximity of Settime to Montechiaro, together with its people's habit to go there [to Montechiaro] for the weekly market creates a relationship that does not exist, and cannot be created, between Settime and Baldichieri. Speaking of which, a consideration is useful. When organising the districts, the goals of the government were: to make the administration of justice easy, to facilitate prompt access to the court and with the least possible inconvenience and the lowest cost to the claimants. The opposite happens when: 1) the town is very far from the district capital; 2) there is no commercial relationship between the town and the district capital; 3) no direct road connects the two towns; 4) there are no adequate bridges on the waterways between the two towns. Therefore, I observe that, if there are some districts composed of 5, 10, or even 20 and more towns and others with 4,3 or even just 2 towns, without any comparability in population, the governments found it useful to organise the districts in such a way. As a consequence, if it would be useful for a town to be part of one district instead of another, then any delay in re-allocating the town to its most convenient district would cause great damage, as the organisation of districts should consider more important the commercial and agrarian or statistical relationships between towns than any misinterpreted consideration about the symmetry, centrality, population size or else; as the district under analysis here is not the only one in such a situation in the Kingdom, I thought it was useful to write a small digression about it."

D Balancedness of observable characteristics



Figure D.1: Balancedness of observable characteristics in the OLS analysis

Notes: The panels of this figure show β coefficients and 95% confidence intervals of separate OLS regressions based under different sample selections. In each panel, units of observation are towns; the explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period. Panel A includes all the town of Piedmont where, in 1899, was born at least one man appearing in the Turin sample (i.e. living in the province of Turin in 1917); Panel B excludes former tax districts capitals from the previous sample; Panel C follows the criteria explained in section 4.3 and excludes all the towns for which the tax district capital hosted a division capital or the closest police station. Panel D, E and F implement the same sample restrictions of A, B and C but keep in the sample only towns of the province of Turin. Dependent variables of the first set of regressions of each panel (in blue) are geographical or pre-determined characteristics, namely: population of the town in 1824; logarithm of elevation of the town; distance from province capital, distance from division capital (*Circondario*), distance from Turin; distances from the Italian borders with Austria, France and Switzerland. Dependent variables of the second set of regressions (in red) are other town-level characteristics, namely: population in 1901 and indicator variables for the presence in the town of a post office, a train station, or a police station.

Table D.1: Balance of observables, geographic and pre-detetermined characteristics

	Pop. 1824	Elevation	Distance Province Cap.	Dist. Turin	Distance Division Cap.	Dist. Austria	Dist. France	Dist. Switzerland
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
(log) Walking Dist. from Tax Collector	0.065	0.024	0.030	0.018	0.372**	0.023	-0.033	-0.018
	(0.154)	(0.063)	(0.059)	(0.053)	(0.181)	(0.052)	(0.057)	(0.018)
DV Mean	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Town-pair FE	yes	yes	yes	yes	yes	yes	yes	yes
Province FE	yes	yes	yes	yes	yes	yes	yes	yes
No district capitals	yes	yes	yes	yes	yes	yes	yes	yes
Towns in sample	135	135	135	135	135	135	135	135
Observations	232	232	232	232	232	232	232	232
R-squared	0.548	0.953	0.973	0.978	0.854	0.991	0.989	0.996

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. The sample includes all the towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Geographical and pre-determined characteristics are: population of the town in 1824, the logarithm of elevation of the town; distance from province capital, distance from division capital (*Circondario*), distance from Turin; distances from the Italian borders with Austria, France and Switzerland. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

	Pop. 1901	Out Migration	In Migration	Post Office	Train Station	Police Station
	(1)	(2)	(3)	(4)	(5)	(6)
(log) Walking Dist. from Tax Collector	0.083	-0.196	0.173	0.059	0.219	0.047
	(0.146)	(0.124)	(0.113)	(0.116)	(0.216)	(0.178)
DV Mean	0.00	0.00	0.00	0.00	0.00	0.00
Town-pair FE	yes	yes	yes	yes	yes	yes
Province FE	yes	yes	yes	yes	yes	yes
No district capitals	yes	yes	yes	yes	yes	yes
Towns in sample	135	135	135	135	135	135
Observations	232	232	232	232	232	232
R-squared	0.676	0.804	0.677	0.586	0.601	0.507

Table D.2:	Balance	of	observables,	town-level	characteristics
------------	---------	----	--------------	------------	-----------------

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. The sample includes all the towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Town-level characteristics are: population in 1901; share of internal immigrants living in the town; share of citizens born in the town living elsewhere; indicator variables for the presence in the town of a post office, a train station, or a police station. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

	Avg. Illiterate	Avg. Height	Cohort Size/	Share Under 150cm
			Population	
	(1)	(2)	(3)	(4)
(log) Walking Dist. from Tax Collector	-0.110	-0.241	0.094	0.147
	(0.124)	(0.248)	(0.225)	(0.150)
DV Mean	0.00	0.00	0.00	0.00
Town-pair FE	yes	yes	yes	yes
Province FE	yes	yes	yes	yes
No district capitals	yes	yes	yes	yes
Towns in Sample	135	135	135	135
Observations	232	232	232	232
R-squared	0.565	0.571	0.771	0.535

Table D.3: Balance of observables, military and means of individual-level characteristics

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. The sample includes all the towns in Piedmont selected as explained in section 4.2. The explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Mean of individual characteristics are share of illiterate citizens born in the town and average height for citizens born in the town. The dependent variable in column (3) is the ratio between the number of citizens born in the town that appear in the 1917 lists and the population of the town in 1901; the dependent variable in column (4) is the share of citizens born in a town less than 150cm tall (the threshold for exemption for medical reasons). Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.



Figure D.2: Balancedness of observable characteristics, town-level averages of individual



Notes: This figure shows β coefficients and 95% confidence intervals of separate regressions based on Equation 4.1, which controls for town-pairs and province fixed effects. Units of observation are towns × town-pair. The sample includes all the towns in Piedmont selected as explained in sections 4.2 and 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period. Dependent variables are: share of illiterate citizens born in the town; average height for citizens born in the town; ratio between the number of citizens born in the town that appear in the 1917 lists and the population of the town in 1901; share of citizens born in a town less than 150cm tall (the threshold for exemption for medical reasons).

E Town-level taxation before 1870, Ivrea and Saluzzo

	Full S	Sample	Restricted Sample	
	Municipal Tax Rev.	Municipal Tax Rev.	Municipal Tax Rev	Municipal Tax Rev i.h.s. trans.
	i.h.s. trans.	i.h.s. trans.	i.h.s. trans.	
	(1)	(2)	(3)	(4)
(log) Walking Dist. from Tax Collector	-0.687***	-0.435**	0.047	-1.427***
	(0.235)	(0.168)	(0.074)	(0.466)
p-value	[0.007]	[0.015]	[0.531]	[0.008]
Wild Cluster Bootstrap-t p-value	[0.002]	[0.028]	[0.556]	[0.019]
DV Mean	8.848	8.848	9.169	9.169
Town-pair FE	yes	yes	yes	yes
Province FE	yes	yes	yes	yes
No district capitals	yes	yes	yes	yes
Geographical controls	no	yes	no	yes
Population	no	yes	no	yes
Towns in Sample	120	120	36	36
Observations	412	412	74	74
R-squared	0.703	0.807	0.747	0.972

Table E.1: Distance from tax collector's residences and local tax revenues, 1831/1846

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns \times town-pair. In every column the dependent variable is the inverse hyperbolic sine transformation of total taxes raised in the town by the tax collector on behalf of the town government, in 1846 (in Ivrea) or in 1831-34 (Saluzzo). In columns (1) and (2), the sample includes all the towns in Piedmont not hosting a former tax collector's residence for which data on local taxes before 1870 are available; in columns (3) and (4), the sample includes all the towns selected as explained in section 4.3 for which data on local taxes before 1870 are available. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Geographical controls and pre-determined controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (Circondario), distance from Turin, population in 1824. Standard errors in round parentheses are clustered at the tax district and district-border level. Square brackets indicate p-values from baseline specification and from wild cluster bootstrap (clustering in both cases at the tax district and district-border level). *** p<0.01, ** p<0.05, * p<0.1.

Figure E.1: Distance from tax collector and municipal tax revenues per capita, Ivrea-Saluzzo sample



Notes: This figure presents graphical evidence on the relationship between the (log) walking distance of a town from a former tax collector's residence in the 1814-1870 period (computed as explained in 3.1) and the amount of taxes per capita raised on behalf of the town government before 1870, transformed using the inverse hyperbolic sine function. Tax data are from 1846 and 1831-1834, and pertain to towns of the divisions of Ivrea (1846) and Saluzzo(1831-34), selected as explained in section 4.3. The binscatter shows the association between the two variables after controlling for town-pair fixed effects, province fixed effects, and geographical and pre-determined characteristics (see B.3 for their descriptions).

Table E.2: Distance from tax collector's residences and per capita local tax revenues,1831/1846

	Full Sample		Restricted Sample		
	Municipal Tax Rev. per Capita				
	i.h.s. trans.	i.h.s. trans.	i.h.s. trans.	i.h.s. trans.	
	(1)	(2)	(3)	(4)	
(log) Walking Dist. from Tax Collector	-0.342	-0.388**	0.076	-1.469**	
	(0.203)	(0.163)	(0.122)	(0.528)	
p-value	[0.103]	[0.025]	[0.546]	[0.014]	
Wild Cluster Bootstrap-t p-value	[0.092]	[0.051]	[0.641]	[0.066]	
DV Mean	8.540	8.540	9.145	9.145	
Town-pair FE	yes	yes	yes	yes	
Province FE	yes	yes	yes	yes	
No district capitals	yes	yes	yes	yes	
Geographical controls	no	yes	no	yes	
Towns in Sample	120	120	36	36	
Observations	412	412	74	74	
R-squared	0.744	0.785	0.734	0.969	

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. In every column the dependent variable is the inverse hyperbolic sine transformation of taxes per capita raised in the town by the tax collector on behalf of the town government, in 1846 (in Ivrea) or in 1831-34 (Saluzzo). In columns (1) and (2), the sample includes all the towns in Piedmont not hosting a former tax collector's residence for which data on local taxes before 1870 are available; in columns (3) and (4), the sample includes all the towns selected as explained in section 4.3 for which data on local taxes before 1870 are available; in columns (3) and (4), the sample includes all the towns selected as explained in section 4.3 for which data on local taxes before 1870 are available. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Geographical controls and pre-determined controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, population in 1824, distance from province capital, distance from division capital (*Circondario*), distance from Turin. Standard errors in parentheses are clustered at the tax district and district-border level. Square brackets indicate p-values from baseline specification and wild cluster bootstrap (clustering in both cases at the tax district and district-border level). *** p<0.01, ** p<0.05, * p<0.1.

Figure E.2: Distance from tax collector and Tax Revenues, before and after the removal of tax districts - years fixed effects



Notes: The figure presents alternative graphical evidence on the relationship between (log) walking distance of a town from a former tax collector's residence in 1814-1870 and the amount of taxes raised by town governments both before and after the removal of tax districts in 1870. The outcome variable amount of taxes raised by town governments before 1870, transformed using the inverse hyperbolic sine function. Plotted coefficients and 95% confidence intervals are estimated using equation 7.1, but controlling for year fixed effects, geographical and pre-determined characteristics (see Table 3 for their description), town-pairs and province fixed effects, and their interaction with year fixed effects. The coefficients plotted in blue are estimated using data from towns of the divisions of Ivrea and Saluzzo for which tax data before 1870 are available, while coefficients plotted in red are estimated using data from all the towns of the Turin sample; for both samples towns are selected as explained in sections 4.2 and 4.3.

	Ivrea and Saluzzo Sample		Main Estimation Sample	
	Municipal Tax Rev	Municipal Tax Rev	Municipal Tax Rev	Municipal Tax Rev
	i.h.s. trans.	i.h.s. trans.	i.h.s. trans.	i.h.s. trans.
	(1)	(2)	(3)	(4)
(log) Walking Distance from Tax Collector \times 1846	-1.427***	-1.427***		
	(0.438)	(0.465)		
(log) Walking Distance from Tax Collector \times 1884	-0.291**	-0.623***	-0.099	-0.087
	(0.123)	(0.186)	(0.071)	(0.078)
(log) Walking Distance from Tax Collector \times 1895	-0.436**	-0.089	-0.155***	-0.164***
	(0.155)	(0.108)	(0.053)	(0.054)
DV Mean	11.86	11.86	12.21	12.21
Town-pair FE	yes	yes	yes	yes
Province FE	yes	yes	yes	yes
No district capitals	yes	yes	yes	yes
Pre/Post FE	yes	no	yes	no
Year FE	no	yes	no	yes
Geographical controls	yes	ys	yes	yes
Years in sample	3	3	2	2
Towns in sample	36	36	135	135
Observations	222	222	464	464
R-squared	0.958	0.982	0.886	0.916

Table E.3: Distance from tax collector and tax revenue in 1831/1846, 1884 and 1895

Notes: The table reports coefficients β_1 , β_2 and β_3 from Equation 7.1. Units of observation are towns \times town-pair × tax year. Every specification includes geographical controls, town-pair and province fixed effects. Columns (1) and (3) include tax year fixed effects and the interaction of a post-reform indicator variable (for the two years after the removal of tax districts in 1870) with every control variable; columns (2) and (4) include tax year fixed effects and their interactions with the every control variable. In columns (1) and (2) the sample includes all the towns selected as explained in section 4.3 for which data on local taxes before 1870 are available; in columns (3) and (4) the sample includes all the towns in Piedmont selected as explained in section 4.3. In every column the dependent variable is the inverse hyperbolic sine transformation of total taxes raised by of the town government in a year. β_1 estimates the effect of distance from a tax district capital in the 1814-1870 period on average yearly taxes collected in the town between 1831 and 1834 (for towns in Saluzzo) or the 1846 tax year (for towns in Ivrea); β_2 and β_3 estimate the same effect for the 1884 and the 1895 tax years. Geographical controls and pre-determined controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, population in 1824, distance from province capital, distance from division capital (Circondario), distance from Turin. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

E.1 Magnitude and validity of the results

In order to make sense of the magnitude of my results, I compare them to previous research that analysed, in different settings, similar outcome variables, or similar phenomena (i.e. the cultural or social consequences of fiscal/state capacity).

First, I show that the magnitude of my results is smaller when compared to previous research that analysed drivers of draft evasion. Rouanet and Piano (2023) study conscription and draft evasion in Napoleonic France, and find how an increase from the first to the third quartile in the distribution of log ruggedness leads to a 27%-59% increase in draft evasion, compared to the mean. While sizeable, my results are smaller: the same increase in log distance from a tax collector leads to a 19,2%-47,1% increase in draft evasion; despite the relevance of cultural motives in draft evasion, it makes sense that geographic characteristics of towns are more relevant for the decision to evade.

Focusing instead on similar independent variables, the reaction of citizens to taxation has been analysed in the Italian setting by Lax-Martinez et al. (2022), who find how towns whose citizens were required to pay a property tax were up to *four times* less likely to have episodes of social unrest in them, compared to towns just below the taxation threshold; using a binary treatment in my analysis based on the median distance from former tax collector's residences I find smaller effects, namely that towns of my sample closer to such a residence have 2-3.3 percentage points fewer evaders born in it in 1899, a 42-70% reduction compared to the mean of 4,72%. The effect drops to a 32-34% reduction when converting my treatment variable into a binary indicator according to whether the town, within a town-pair, is farther or closer to the tax collector (as in Table H.2).

Very interestingly, the magnitude of my results can also be compared to findings of previous research that studies the long run effects of state capacity on culture (and norms of compliance more specifically) in sub-Saharan Africa: indeed, Heldring (2021) finds that an increase in a proxy for state capacity (i.e. years of state presence) equal to its mean leads to an increase in state-sponsored violence against minorities equal to its mean: at the mean, decreasing my measure of historical fiscal capacity by its mean (i.e. a 100% increase in distance) leads to a 55% increase in draft evasion. My result, while large and comparable in size, is therefore clearly smaller than the one found by Heldring (2021).

Also in the European setting, researchers have studied the long run effects of state capacity on norms of compliance, finding quantitatively large results; for instance, Heldring (2023) finds how German towns previously under Prussia (i.e. with higher state capacity) were 44% more effective when asked to deport Jews from the Nazi government; once again, such a result is comparable in magnitude to the ones I estimate (42-70% or 32-34%, according to the method used to convert my treatment in discrete).

As my treatment is continuous while related papers use a discrete one, another way of making sense of the magnitude of my effect is to convert other papers' effects into kilometres that would be needed in my setting to reach such an effect size. To obtain an effect of the same magnitude as Lax-Martinez et al. (2022) (which is a 67% reduction in social unrest

in the baseline specification), at the sample mean, I would need to increase distance of a town from a former tax collector's residence by 8,93 kilometres, an increase larger than mean distance and more than twice the interquartile range in distance. Regarding Heldring (2023), an effect of similar magnitude would be reached by increasing distance from tax collector by 5,87 kilometres, almost twice the interquartile range.

One likely explanation for the generally smaller magnitude of my results compared to the aforementioned studies is my focus on institutions for fiscal capacity, instead of a bundle of institutions, including legal capacity. Indeed, exposure to other institutions likely produced additional non-negligible effects on citizens' culture; as such effects on cultural norms could be going in the same direction as the effect of fiscal capacity, isolating the latter plausibly leads me to estimate smaller effects.

F Additional results on Vicenza sample

	Share Draft Evaders			
	(1)	(2)	(3)	
(log) Walking Dist. from District Capital	-0.000	0.016	0.007	
(105) Walking Dist. Hom District Capital	(0.031)	(0.021)	(0.019)	
p-value	[0.994]	[0.475]	[0.735]	
Wild Cluster Bootstrap-t p-value	[0.993]	[0.597]	[0.789]	
DV Mean	0.0551	0.0551	0.0551	
Town-pair FE	yes	yes	yes	
Province FE	yes	yes	yes	
No district capitals	yes	yes	yes	
Geographical controls	no	yes	yes	
Mean individual controls	no	no	yes	
Town-level controls	no	no	yes	
Observations	3,500	3,500	3,500	
Individuals in Sample	1799	1799	1799	
Towns in Sample	55	55	55	
R-squared	0.031	0.038	0.040	

Table F.1: Distance from judicial district capitals and likelihood to evade the draft

Notes: The table reports β coefficients from Equation 4.2, which controls for town-pair and province fixed effects. Units of observation are individuals \times town-pair. In every column the dependent variable is a binary variable indicating whether the individual evaded the draft, or not. The sample includes all the individuals born in towns out of Piedmont selected as explained in section 4.3 and 6.3. The main explanatory variable is the logarithm of the walking distance between a town and the location of a minor court that had jurisdiction on the town (or district capital), computed as explained in section 3.1. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (Distretto), distance from Turin, imputed population in 1843-44. Mean of individual controls are town-level means of individual characteristics as measured by military data, namely being resident abroad and having had an exemption for medical reasons. Town-level controls include population in 1901, distance from the closest police station, and indicator variables for the presence in the town of a post office, a police station, or a train station. Standard errors in parentheses are clustered at the tax district and district-border level. Square brackets indicate p-values from baseline specification and from wild cluster bootstrap (clustering in both cases at the tax district and district-border level). *** p<0.01, ** p<0.05, * p<0.1.
	Regression sample				Full sample					
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Obs.	Mean	Std. Dev.	Min.	Max.	Obs.	Mean	Std. Dev.	Min.	Max.
Share of draft evaders	55	0.04	0.05	0.00	0.25	201	0.03	0.08	0.00	1.00
Walking dist. from tax collector's residence (metres)	55	9532.11	3388.21	4371.00	19392.00	199	8660.25	20760.49	0.00	292882.00
log walking dist. from tax collector's residence	55	9.11	0.33	8.38	9.87	165	9.02	0.49	7.61	12.59
log Elevation	55	5.00	1.42	2.40	7.03	201	4.47	1.58	-0.59	7.42
log Area	55	2.95	0.58	1.09	4.10	201	3.14	0.75	1.09	6.03
Latitude	55	45.59	0.18	45.23	45.94	201	45.54	0.31	44.23	46.48
Longitude	55	11.48	0.20	11.19	11.85	201	11.59	0.29	10.47	12.69
Dist. from Division Capital (km)	55	23.45	12.36	5.22	64.27	194	18.70	12.85	0.00	64.27
Dist. from Province Capital (km)	55	32.55	12.65	16.59	68.77	194	28.05	16.59	0.00	88.80
Dist. from Turin (km)	55	391.97	51.04	314.30	467.65	197	398.24	53.46	251.40	561.98
Dist. from Austrian Border (km)	55	27.69	15.43	2.70	62.99	201	38.92	28.17	1.54	178.15
Dist. from French Border (km)	55	334.27	16.24	309.81	365.76	201	340.94	23.92	230.71	432.68
Dist. from Swiss Border (km)	55	126.48	15.34	98.69	158.91	201	137.99	26.99	95.32	268.69
Population 1901	55	3428.20	1373.16	633	6591	201	6447.45	14596.51	633	160096
Presence of train station	55	0.15	0.36	0.00	1.00	201	0.23	0.42	0.00	1.00
Presence of post office	55	0.82	0.39	0.00	1.00	201	0.74	0.44	0.00	1.00
Presence of police station	55	0.29	0.46	0.00	1.00	201	0.32	0.47	0.00	1.00
Distance from police station (metres)	55	3765.76	2756.11	0.00	9326.00	199	3715.71	2888.40	0.00	10071.00
Out migration	55	4.20	3.22	0.00	17.00	201	5.53	6.26	0.00	58.00
In migration	55	3.29	4.23	0.00	15.00	201	5.52	15.58	0.00	202.00
Observations per town (town-pairs)	55	1.82	1.04	1.00	5.00	201	4.63	2.32	1.00	12.00
Share exempted citizens	55	0.07	0.14	0.00	1.00	201	0.06	0.12	0.00	1.00
Average share of migrants abroad	55	0.01	0.02	0.00	0.13	201	0.01	0.02	0.00	0.20

 Table F.2:
 Summary statistics:
 Vicenza sample

Notes: The table presents summary statistics for town-level variables used in the placebo analysis. Variables' construction is described in appendix B.3. Columns (1) to (5) present summary statistics from towns of the Vicenza regression sample, selected as explained in sections 4.3 and 6.3, and individuals living in these towns. Columns (6) to (10) present summary statistics from all the non-Piedmontese towns of birth of the drafted citizens in the Vicenza sample.

G Additional results on mechanisms

G.1 Turnout in 1913 national elections - variable's construction

I compute turnout in 1913 elections as the ratio between total votes expressed in 1913 and *de facto* eligible voters in 1913. I use data from Corbetta and Piretti (2009) and the historical archive of the Italian Chamber of Deputies to obtain total votes expressed in 1913 in each precinct; I then aggregate such information at the town level.

To build a measure of the *de facto* franchise, I start with precinct-level information on eligible voters collected from the historical archive of the Italian Chamber of Deputies, which I call de *iure* eligible voters, and aggregate them at the town level. However, the number of eligible voters recorded by the Italian Chamber of Deputies, per the 1912 electoral law,⁷⁰ included also permanent Italian migrants living abroad, and men serving in the Army at the time of the elections, who were forbidden to vote. As a consequence, the number of *de iure* eligible voters (the one recorded by the Italian Chamber of Deputies) includes citizens incapacitated to vote in 1913, which is likely to lead to a distorted measure of voter turnout. While town-level information on the number of citizens incapacitated to vote because residents abroad or serving in the Army is not available, Direzione Generale della Statistica (1913) contains information on both de iure and de facto eligible voters at the electoral district level, a geographical unit including several towns (and precincts); I define de facto eligible voters as all the voters who are not classified as permanent migrants abroad or serving in the Army during the election. I use such information from Direzione Generale della Statistica (1913) to build a town-level measure of *de facto* eligible voters proceeding as follows: after verifying the correspondence between the number of de iure voters in the Italian Chamber of Deputies' data and Direzione Generale della Statistica (1913), I compute an electoral district-level correction coefficient as the ratio between de facto and de iure eligible voters, and multiply it times the number of *de iure* eligible voters in each town to obtain my town-level measure of the *de facto* franchise. Finally, I build my measure of turnout in the 1913 national elections as:

$$Voter National Turnout_{te} = \frac{Total Votes_{te}}{De Jure Voters_t \times \frac{De Facto Voters_e}{De Jure Voters_e}},$$
(G.1)

where t represents the town and e the electoral district the town belongs to.

In a limited number of instances, precincts are larger than towns, and include several of them; as a consequence, it is not possible to obtain town-level data for such observations. In these cases, I aggregate all the towns belonging to the same precinct into one unique data point.

 $^{^{70}}Legge \ 30 \ giugno \ 1912, \ n. \ 666, articles \ 15 \ and \ 29.$

	Local Electi	ons Franchise	National Elections Franchise			
	18	865		1913		
	(1)	(2)	(3)	(4)		
(log) Walking Dist. from Tax Collector	-0.020	-0.000	-0.002	0.003		
	(0.014)	(0.012)	(0.006)	(0.011)		
DV Mean	0.396	0.396	0.675	0.675		
Town-pair FE	yes	yes	yes	yes		
Province FE	yes	yes	yes	yes		
No district capitals	yes	yes	yes	yes		
Geographical controls	no	yes	no	yes		
Towns in sample	135	135	135	135		
Observations	232	232	232	232		
R-squared	0.677	0.828	0.631	0.703		

Table G.1: Distance from former tax district capitals and electoral franchise, 1865 and1913

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. The sample includes all the towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. In columns (1) and (2) the dependent variable is the electoral franchise in 1865, computed using data from Statistica del Regno d'Italia (1867) as the share of male citizens entitled to vote in local elections in 1865 out of the town's population in 1862 . In columns (3) and (4) the dependent variable is the electoral franchise in 1913, computed using data from the historical archive of the Italian Chamber of Deputies and Direzione Generale della Statistica (1913) as the share of male citizens entitled to vote in national elections in 1913 out of the town's population in 1901. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distance from Turin. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.



Figure G.1: Turnout in local and national elections: variables' distributions

Notes: The two histograms represent the distribution of the two measures of voter turnout (in local 1865 elections and 1913 elections) in the towns of the estimation sample, selected as explained in section 4.3.

H Robustness checks

H.1 Alternative empirical strategy

 Table H.1: Distance from former tax district capitals and share of draft evaders: Border fixed-effects design

	Share Draft Evaders				
	(1)	(2)	(3)		
(log) Walking Dist. from Former Tax Collector	0.006	0.019**	0.025*		
(0)	(0.009)	(0.009)	(0.013)		
DV Mean	0.0453	0.0453	0.0457		
Border FE	yes	yes	yes		
Province FE	yes	yes	yes		
No district capitals	yes	yes	yes		
Geographical controls	no	yes	yes		
Mean individual controls	no	no	yes		
Town-level Controls	no	no	yes		
Towns in Sample	155	155	153		
Observations	234	234	232		
R-squared	0.442	0.518	0.595		

Notes: The table reports β coefficients from Equation 6.1, which controls for district-border and province fixed effects. Units of observation are towns \times town-pair. In every column the dependent variable is the share of draft evaders born in a town in 1899 out of all the men born in 1899 in the town. The sample includes all the towns in Piedmont which shared an administrative boundary with a district-border, which respect the sample selection explained in section 4.3 and located on a district-border with at least one town on each side of it. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (Circondario), distance from Turin, population in 1824. Mean of individual controls are town-level means of individual characteristics as measured by military data, including height, literacy, being resident abroad, and having had an exemption for medical reasons. Town-level controls include population in 1901, distance from the closest police station, and indicator variables for the presence in the town of a post office, a police station, or a train station. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

H.2 Alternative measures of distances from tax collectors' residences

	Share Draft Evaders							
	(1)	(2)	(3)	(4)	(5)	(6)		
(log) Euclidean Dist. from Former Tax Collector	0.015^{**}	0.026^{**}						
	(0.007)	(0.012)						
(log) Cost Distance from Tax Collector			0.014^{**}	0.023^{*}				
			(0.006)	(0.012)				
Treatment (Farther from Tax Collector)					0.015	0.016		
					(0.010)	(0.013)		
DV Mean	0.0472	0.0472	0.0472	0.0472	0.0472	0.0472		
Town-pair FE	yes	yes	yes	yes	yes	yes		
Province FE	yes	yes	yes	yes	yes	yes		
No district capitals	yes	yes	yes	yes	yes	yes		
Geographical controls	no	yes	no	yes	no	yes		
Towns in Sample	135	135	135	135	135	135		
Observations	232	232	232	232	232	232		
R-squared	0.587	0.678	0.586	0.676	0.591	0.676		

 Table H.2: Distance from former tax district capitals and share of draft evaders:

 Alternative measures of distance

Notes: The table reports coefficients β from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. In every column the dependent variable is the share of draft evaders born in a town in 1899 out of all the men born in 1899 in the town. The sample includes all the towns in Piedmont selected as explained in section 4.3. In columns (1) and (2) the main explanatory variable is the logarithm of the euclidean distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital); in columns (3) and (4) the main explanatory variable is the logarithm of the cost distance (in days) between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed using the Human Mobility Index by Özak (2010); in columns (5) and (6) the main explanatory variable is a binary variable indicating whether the town, within the couple, is farther from the residence of its tax collector's residence, according to the walking distance computed as described in section 3.1. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distance from Turin, population in 1824. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

H.3 Alternative measure of towns' draft evasion

		Any Draft Evader	
	(1)	(2)	(3)
(log) Welling Dist from Tay Collector	0 191	0.179*	0 096***
(log) waiking Dist. from Tax Conector	(0.087)	(0.095)	(0.011)
DV Mean	0 499	0.422	0 422
DV mean	0.422	0.422	0.422
Town-pair FE	yes	yes	yes
Province FE	yes	yes	yes
No district capitals	yes	yes	yes
Geographical controls	no	yes	yes
Mean individual controls	no	no	yes
Town-level controls	no	no	yes
Towns in sample	135	135	135
Observations	232	232	232
R-squared	0.659	0.802	0.776

Table H.3: Distance from former tax district capitals and binary outcome: any draftevader born in town

Notes: The table reports coefficients β from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns × town-pair. In every column the dependent variable is a binary variable taking value 1 for towns with at least one draft evader born in them in 1899, and 0 otherwise. The sample includes all the towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distance from country borders, distance from province capital, distance from division capital (*Circondario*), distance from Turin, population in 1824. Mean of individual controls are town-level means of individual characteristics as measured by military data, including height, literacy, being resident abroad, and having had an exemption for medical reasons. Town-level controls include population in 1901, distance from the closest police station, and indicator variables for the presence in the town of a post office, a police station, or a train station. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

H.4 Alternative sample restrictions

 Table H.4: Distance from former tax district capitals and share of draft evaders,

 Alternative estimation samples

	Only closest matched town		Excluding province capital's district		Division fixed effects		Including tax-district capitals		Only Province of Turin	
	Share Draft Evaders	Share Draft Evaders	Share Draft Evaders	Share Draft Evaders	Share Draft Evaders	Share Draft Evaders	Share Draft Evaders	Share Draft Evaders	Share Draft Evaders	Share Draft Evaders
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
(log) Walking Dist. from Former Tax Collector	0.020*	0.024	0.011*	0.022**	0.016**	0.031**			0.015**	0.031**
	(0.011)	(0.016)	(0.007)	(0.009)	(0.006)	(0.014)			(0.006)	(0.013)
Walking Dist. from Tax Collector, i.h.s. trans.							0.001 (0.002)	0.003 (0.002)		
DV Mean	0.0453	0.0453	0.0426	0.0426	0.0485	0.0485	0.0471	0.0471	0.0538	0.0538
Town-pair FE	yes	yes	yes	yes	yes	yes	yes	yes		
Province FE	yes	yes	yes	yes	no	no	yes	yes	yes	yes
Division FE	no	no	no	no	yes	yes	no	no	no	no
No district capitals	yes	yes	yes	yes	yes	yes	no	no	yes	yes
Geographical controls	no	yes	no	yes	no	yes	no	yes	no	yes
Towns in Sample	135	135	161	161	130	130	144	144	99	99
Observations	180	180	286	286	226	226	270	270	188	188
R-squared	0.564	0.667	0.591	0.676	0.586	0.679	0.570	0.664	0.592	0.745

Notes: The table reports coefficients β from Equation 4.1, controlling for town-pair fixed effects. Units of observation are towns \times town-pair. In every column the dependent variable is the share of draft evaders born in a town in 1899 out of all the men born in 1899 in the town. In columns (1) and (2) the sample includes the towns in Piedmont selected as explained in section 4.3; however, instead of using all the pairs of neighbouring towns on tax district borders, for each town I keep only the pair with the geographically closest matched town. In columns (3) and (4) I exclude towns of the province capital tax district, instead of towns of the division capital tax district. In columns (5) and (6) I control for division fixed effects instead of province fixed effects. In columns (7) and (8) I keep in the estimation sample also tax district capitals that respect the sample selection's criteria of 4.3. In columns (9) and (10) I restrict my analysis only to towns in the province of Turin. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (Circondario), distance from Turin, population in 1824. Standard errors in parentheses are clustered at the tax district and district-border level. *** p<0.01, ** p<0.05, * p<0.1.

H.5 Alternative clustering

		rs	
	(1)	(2)	(3)
(log) Walking Dist. from Tax Collector	0.015**	0.026**	0.036***
[5 km]	(0.006)	(0.007)	(0.004)
$[10 \ km]$	(0.006)	(0.005)	(0.003)
$[20 \ km]$	(0.006)	(0.003)	(0.002)
$[50 \ km]$	(0.007)	(0.002)	(0.001)
[100 km]	(0.005)	(0.002)	(0.001)
DV Mean	0.0472	0.0472	0.0472
Town-pair FE	yes	yes	yes
Province FE	yes	yes	yes
No district capitals	yes	yes	yes
Geographical controls	no	yes	yes
Mean individual controls	no	no	yes
Town-level controls	no	no	yes
Towns in sample	135	135	135
Observations	232	232	232
R-squared	0.587	0.678	0.776

Table H.5: Distance from former tax district capitals and share of draft evaders: SpatialHAC standard errors

Notes: The table reports β coefficients from Equation 4.1, which controls for town-pair and province fixed effects. Units of observation are towns \times town-pair. In every column the dependent variable is the share of draft evaders born in a town in 1899 out of all the men born in 1899 in the town. The sample includes all the towns in Piedmont selected as explained in section 4.3. The main explanatory variable is the logarithm of the walking distance between a town and the residence of its tax collector in the 1814-1870 period (or district capital), computed as explained in section 3.1. Geographical controls include the logarithm of elevation of the town, standard deviation of elevation in the town, area of the town, measures of land suitability for several crops (as explained in Appendix B.3), latitude, longitude, distances from country borders, distance from province capital, distance from division capital (Circondario), distance from Turin, population in 1824. Mean of individual controls are town-level means of individual characteristics as measured by military data, including height, literacy, being resident abroad, and having had an exemption for medical reasons. Townlevel controls include population in 1901, distance from the closest police station, and indicator variables for the presence in the town of a post office, a police station, or a train station. Conley's spatially-corrected standard errors in parentheses. Following the methodology introduced by Colella et al. (2019), I allow errors to be correlated within a circle of radius of 5, 10, 20, 50 or 100 kilometres around each town. *** p<0.01, ** p<0.05, * p<0.1.

H.6 Role of unobservables: Oster test



Figure H.1: Oster's δ under alternative assumptions about R_{max}^2 cutoff

Notes: This figure presents the value of Oster's δ for several values of R_{max}^2 , where R_{max}^2 ranges between $1.1\bar{R}^2$ and $1.6\bar{R}^2$ according to the different cutoffs. \bar{R}^2 is the R-squared from the baseline regression, controlling for province and town-pairs fixed effects only. The δ statistic indicates how much more important unobservables need to be compared to observables to have omitted variable bias fully explaining the results. The dashed vertical line denotes 1.3, the cutoff suggested by Oster (2019). The solid horizontal line denotes $\delta = 1$, which indicates equal selection on observables and unobservables.